







Networks of Centres of Excellence Réseaux de centres d'excellence



Scientific theory, data and techniques for conservation of water resources within a changing forested landscape

Irena Creed | Gabor Sass | Fred Beall | Jim Buttle | Dan Moore | Margaret Donnelly

A STATE OF KNOWLEDGE REPORT















Networks of Centres of Excellence Réseaux de centres d'excellence

A STATE OF KNOWLEDGE REPORT

Scientific theory, data and techniques for conservation of water resources within a changing forested landscape

Irena Creed, Department of Biology, University of Western Ontario Gabor Sass, Department of Biology, University of Western Ontario Fred Beall, Great Lakes Forestry Centre, Canadian Forest Service Jim Buttle, Department of Geography, Trent University Dan Moore, Department of Geography, University of British Columbia Margaret Donnelly, Alberta-Pacific Forest Industries, Inc.

© 2011, Sustainable Forest Management Network*

*F. Beall © 2011 Her Majesty the Queen in right of Canada, Natural Resources Canada / Canadian Forest Service and Sustainable Forest Management Network

This publication contains sections from the Springer Verlag book *"Forest Hydrology and Biogeochemistry - Synthesis of Past Research and Future Directions"*. No part of Springer's work may be reproduced, stored in a retrieval system, or transmitted in any form or by any means, electronic, mechanical, photocopying, microfilming, recording or otherwise, without written permission from the Springer Verlag, with the exception of any material supplied specifically for the purpose of being entered and executed on a computer system, for exclusive use by the purchaser of the work.

The sections of this publication that are not taken from the Springer Verlag book *"Forest Hydrology and Biogeochemistry - Synthesis of Past Research and Future Directions"* may be reproduced in whole or in part for non-commercial use without permission provided that its source is fully acknowledged. Reproduction of this publication in whole or in part for any other purpose, including commercial sale or distribution, requires prior written permission from the Sustainable Forest Management Network.

No guarantee or warranty, expressed or implied, is made about the value or stability of the information or links made herein.

The views, conclusions and recommendations contained in this publication are those of the authors and should not be construed as endorsement by the Sustainable Forest Management Network.

Citation: Creed, I. F., Sass, G. Z., Beall, F. D., Buttle, J. M., Moore, R. D. and Donnelly, M. 2011. Scientific theory, data and techniques for conservation of water resources within a changing forested landscape. A State of Knowledge report. Sustainable Forest Management Network, Edmonton, Alberta. 136 pp.

(See also the companion report prepared by Creed et al. 2011. *Hydrological principles for conservation of water resources within a changing forested landscape*. A State of Knowledge report. 80 pp.)

For an electronic version of this report, visit the Sustainable Forest Management Network legacy website at www.ales.ualberta.ca/forestry/Sustainable_Forest_Management.aspx.

Print copies are available free of charge while supplies last.

Library and Archives Canada Cataloguing in Publication

Scientific theory, data and techniques for conservation of water resources within a changing forested landscape / Irena Creed ... [et al.].

Includes bibliographical references and index. Type of computer file: Electronic monograph in PDF format. Issued also in print format. ISBN 978-1-55261-276-71

1. Water conservation--Canada. 2. Forest management--Canada. 3. Forest hydrology--Canada. 4. Sustainable forestry--Canada. 5. Forest policy--Canada. I. Creed, Irena Frances, 1961-

TD226.S35 2011a 333.9100971 C2011-905263-6

Photography	Design
Front Cover (top to bottom):	www.c3design.ca
 SFMN archives 	
• Stephen Kaufman	Printing
• Bruce Macnab	Priority Printing Ltd.
Background:	Printed in Canada
 SFMN archives 	Published October 2011
Back Cover: • Scott Brown	

Foreword

The State of Knowledge program was launched by the Sustainable Forest Management Network (SFMN) to capture the knowledge and wisdom that had accumulated in publications and people over a decade of research. The goal was to create a foundation of current knowledge on which to build policy, practice and future research. The program supported groups of researchers, working with experts from SFMN partner organizations, to review literature and collect expert opinion about issues of importance to Canadian forest management. The priority topics for the program were suggested by the Network's partners in consultation with the research theme leaders. Each State of Knowledge team chose an approach appropriate to the topic. The projects involved a diversity of workshops, consultations, reviews of published and unpublished materials, synthesis and writing activities. The result is a suite of reports that we hope will inform new policy and practice and help direct future research.

The State of Knowledge program has been a clear demonstration of the challenges involved in producing a review that does justice to the published literature and captures the wisdom of experts to point to the future. We take this opportunity to acknowledge with gratitude the investment of time and talent by many researchers, authors, editors, reviewers and the publication production team in bringing the program to a successful conclusion.

J.a. Wityh

Jim Fyles Scientific Director

Mu Chum

Fraser Dunn Chair of the Board

Acknowledgements

4

We gratefully acknowledge funding from the Sustainability Forest Management Network. Additional funding and support was generously provided by a Canada Research Chair grant (Creed), a NSERC Discovery Grant (Creed) and an NSERC Postdoctoral Fellowship (Sass).

We would like to thank our SFM Network supporting partners who identified information pertaining to the conservation and management of water resources as a critical issue in sustainable forest management which led to the idea for this report. In particular, we would like to acknowledge the input of the following project partners who attended workshops, contributed ideas and questions related to water and forest management, provided resource materials and guidance on report content: Abitibi-Consolidated Inc.; Alberta-Pacific Forest Industries Inc.; Government of Alberta (Sustainable Resources Department); Ducks Unlimited Canada Inc.; Environment Canada; LP Canada Ltd.; Moose Cree First Nation; Natural Resources Canada – Canadian Forest Service; and Government of Ontario (Ministry of Natural Resources).

We would like to thank Dr. Del Levia (University of Delaware), Dr. Darryl Carlyle-Moses (Thompson Rivers University), and Dr. Tadashi Tanaka (University of Tsukuba), for the invitation to contribute chapters on the effects of forest harvesting on forest hydrology and biogeochemistry, and the use of digital terrain analysis and remote sensing in forest hydrology and biogeochemistry to the recent Springer-Verlag publication titled *Forest Hydrology and Biogeochemistry: Synthesis of Past Research and Future Directions.* Special thanks go to Dr. Del Levia for his passionate encouragement in the preparation of these materials.

We would like to acknowledge Springer-Verlag for providing permission for the inclusion of Chapter 2 ("The Effects of Forest Harvesting on Forest Hydrology and Biogeochemistry"), Chapter 4 ("Bird's Eye View of Forest Hydrology: Novel Approaches using Remote Sensing Techniques"), and "Chapter 5 ("Digital Terrain Analysis Approaches for Tracking Hydrological and Biogeochemical Pathways and Processes in Forested Landscapes") that were published in *Forest Hydrology and Biogeochemistry: Synthesis of Past Research and Future Directions*.

We would like to thank Mr. Johnston Miller (The University of Western Ontario) for editorial comments that improved the quality of this report.

Finally, we would like to thank Dr. Magdalena Burgess (McGill University) and Catalina Solano Rivera (University of Alberta) for transforming the document into the format of the SFMN State of Knowledge series.

Table of Contents

3		Foreword		
4		Acknowledgements		
9		Executive Summary		
17	1.0	Introduction		
21	2.0	The effects of forest harvesting on forest hydrology and biogeochemistry		
21	2.1	Introduction		
21	2.2	Effects of forest harvesting on hydrologic processes and water partitioning		
25	2.3	Effects of forest harvesting on streamflow		
28	2.4	Biogeochemical aspects of forest harvesting		
32	2.5	Issues for future research		
35	3.0	Canada's national hydrological datascape		
35	3.1	Introduction		
36	3.2	Experimental watersheds		
41	3.3	Hydrometric networks: the Water Survey of Canada		
43	3.4	Gaps in Canada's current sources of hydrological data		
45	3.5	Recommendations and conclusions		
47	4.0	Bird's-eye view of forest hydrology: Novel approaches using remote sensing techniques		
47	4.1	Introduction		
48	4.2	A primer for forest hydrologists		
52	4.3	Water budget		
60	4.4	From science to practice		
62	4.5	Towards an operational bird's-eye view		
64	4.6	Conclusions		
65	5.0	Digital terrain analysis approaches for tracking hydrological and biogeochemical pathways and processes in forested landscapes		
65	5.1	Introduction		
67	5.2	Digital terrain analysis for forest hydrologists		
75	5.3	Tracking hydrological pathways using digital terrain analysis		
80	5.4	Tracking biogeochemical pathways using digital terrain analysis		
85	5.5	From science to practice		
86	5.6	Towards an operational digital terrain analysis approach		
88	5.7	Conclusions		

continued on the next page

89	6.0	/irtual ecosystems: Using hydrological models to understand the past and predict the future of water in forests		
89	6.1	Introduction		
91	6.2	How to pick your model: a primer		
101	6.3	Nodels and forest management		
106	6.4	Modelling the future of water resources under a changing climate		
107	6.5	Barriers to the operational use of models		
109	6.6	Conclusions		
111	7.0	Recommendations and conclusions		
115	8.0	References cited		

b

List of Figures

25	Figure 1.	a – Annual water yield increase vs. % of basin harvested, based on experimental studies in the USA; b – Annual water increase vs. basin area for all basins and basins with 100% of their area harvested.
27	Figure 2.	 a – % change in bankfull peak flows vs. % of basin cut or volume of standing timber harvested, based on experimental studies from around the world; b – % change in bankfull peak flows vs. basin area for all basins, basins with 100% of their area cut or volume of standing timber cut, and basins where mean annual precipitation (MAP) exceeds 1200 mm
31	Figure 3.	Hypothesized changes in streamflow concentration (C) relative to maximum concentration (Cmax) for non-essential, essential and limiting chemicals after forest harvesting and throughout forest regrowth
37	Figure 4.	Location of forest hydrology experimental study sites in different ecozones of Canada
40	Figure 5.	Example of metadata sheet compiled for Turkey Lakes Watershed experimental site by HydroEcological Landscapes and Processes (HELP) project
42	Figure 6.	Activity of watershed studies
44	Figure 7.	Frequency distribution of watershed areas for experimental research sites in Canada compared with the WSC hydrological data base
48	Figure 8.	Number of publications per year in forest hydrology
50	Figure 9.	Remote sensors useful for the study of forest hydrology
59	Figure 10.	Spatial and temporal resolution of remotely sensed data with a shaded polygon defining limits of currently measurable resolutions
61	Figure 11.	Adaptive-width buffer design
67	Figure 12.	Conceptual model of runoff generating mechanisms in three different terrains
70	Figure 13.	Hydrologically important terrain attributes used in the modeling of water, nutrient and sediment redistribution along a hillslope
71	Figure 14.	Hydrologically important terrain features used in the modeling of water, nutrient and sediment redistribution along a hillslope
76	Figure 15.	Comparison of digital terrain analysis (DTA) techniques for wet area mapping
84	Figure 16.	Conceptual model of water and nutrient source areas in VSA dominated landscapes
92	Figure 17.	Model evaluation framework for forest hydrologists

List of Tables

19	Table 1.	Hydrological principles of forest management
30	Table 2.	Major factors governing stream chemistry in forest landscapes and suggested concentration response of selected nutrients and contaminants to harvesting effects on these governing factors
38	Table 3.	Basic descriptive information on forest hydrology experimental study sites organized by ecozones
51	Table 4.	Potential application of electromagnetic spectrum to study of forest hydrology
53	Table 5.	Published reviews relevant to remote sensing of forest hydrology
53	Table 6.	Relative error of RS derived water budget components
69	Table 7.	Evolution of hydrological flow routing algorithms
73	Table 8.	Digital terrain indices used to infer different hydrological and biogeochemical patterns and processes in forested landscapes
94	Table 9.	Structural properties of models characterized by spatial and temporal discretization, process representation
96	Table 10.	Model complexity and its relation to data, resource and time requirements for any modeling exercise
98	Table 11.	Models that have been useful in the study of forest hydrology

List of Boxes

20	Box 1.	Terminology: Catchment, watershed, and drainage basin
49	Box 2.	Terms and terminology: Remote sensing techniques

Executive Summary

This State of Knowledge (SOK) report builds on a companion SOK report entitled "*Hydrological principles for conservation of water resources within a changing forested landscape*", which presents a suite of hydrological principles that provide a foundation for the development of sustainable forest management strategies to conserve water resources.

With this report, our objective is to review the state of science resources (including data and tools) behind the sustainable management of forests from the perspective of conserving water resources and minimizing adverse effects resulting from forest management activities. We also provide a current synthesis of field studies and available datasets, as well as the scientific achievements and challenges facing the application of digital tools including digital terrain analysis, remote sensing and hydrological modelling. Finally, we provide recommendations for scientists, policy makers and resources managers with respect to future research and monitoring endeavours, analysis of integrated datasets and training of the next generation of forest hydrologists and forest managers to promote the practice of sustainable forest management.

Where does sustainable forest management come from?

Sustainable forest management strategies for the conservation of water resources are based on the results of hundreds of scientific studies from Canada and abroad. These studies have sought to understand the consequences of management activities on hydrological processes in forested landscapes. In order to do so, research has been conducted at a series of experimental study sites, most being headwater catchments, spanning a diverse range of forested regions. In Canada, there are approximately 50 sites which have produced most of the forest hydrology research.

Do forest management activities change hydrological processes?

They potentially do, but the effects are variable across spatial and temporal scales. Different forest types will respond differently to forest harvesting, and the responses in a given forest will differ over the course of a year. Additionally, global change is altering how we look at the effects of forest harvesting, because it is modifying the fundamental hydrological processes occurring within our forests.

Future research should focus on maintaining and expanding the monitoring network of sites that are examining the long-term hydrological and biogeochemical characteristics of a wide variety of natural and harvested forests. These sites could be used to investigate how we can translate results from experimental catchments to larger regional drainage basins. Specifically, it is important to learn more about how the effects of forest harvesting change with scale and how they compare with other land use/land cover changes. It is also important to look beyond issues of annual water yield and consider other aspects of streamflow regimes including streamflow magnitudes (high and low flows), timing of streamflows, frequencies, and the durations of streamflow events.

The effects of forest harvesting on forest hydrology and biogeochemistry

Water quantity and quality issues, especially as they relate to the provision of safe and sufficient drinking water supply will be increasingly important under a changing climate, particularly since many predict an increased frequency and severity of extreme weather events. For these reasons, it is crucial to have a clear understanding of the impacts of forest harvesting on water. The key questions asked by hydrologists working in managed forests are:

- 1. How do forest management activities alter hydrological processes?; and
- 2. How do we mitigate the impacts?

We now have a fairly good grasp on many of these changes, although knowledge gaps remain, specifically for subsurface (ground water) hydrological processes.

Changes in hydrological processes

In general, basins that have been harvested produce greater water yields. This response, however, is highly variable and is affected by such factors as climate, geology, soils, topography and vegetation. Additionally the effects of forest harvest are not static, and diminish over time as the forest re-grows. Some potential reasons why greater water yields can be expected in harvested areas are:

- Higher snow accumulation, accompanied by more rapid snow melt in the spring due to higher radiation exposure.
- Higher melt rates in regions subject to mid-winter rain-on-snow events, due to an increase in wind-driven inputs of sensible and latent heat to the snow surface.
- Higher soil water content during the snow-free season as a result of fewer trees intercepting rain and snow, and an overall reduction in evapotranspiration due to reductions in leaf area.

Increased amounts of water in the soil in harvested watersheds translate into higher water tables, increased groundwater recharge, and greater groundwater discharge to streams and rivers; although the magnitude of these effects are highly variable. Higher soil water could also lead to increased subsurface flow and change the rate at which water is able to infiltrate into the soil. If infiltration rates have been altered through forestry operations, there can be resulting increases in overland flow. The magnitude of all these effects, however, depends on the techniques used to remove the trees (e.g., soil compaction from equipment, roads).

The issue of increased peak flows due to forest harvesting, and the associated risk of flooding, is a contentious issue in hydrology. However, there is general consensus that forest harvesting has a proportionally larger effect on smaller rather than larger basins, and that effects are generally observed when more than 20% of a basin area is harvested. There is also general consensus that low flows during dry periods increase following harvesting due to increased soil and groundwater recharge, at least for the first five to ten years following harvest. Again, these effects are highly variable and longer-term effects (10 years +) are virtually unknown.

Water quality

Water quality issues are complex and interconnected. In fact, the effects of forest harvesting can be difficult to tease apart from issues such as climate, vegetation and land use history. Also, there are many indicators that represent quality based on different variables and that affect the ecosystem in diverse ways.

Stream temperature is an important water quality indicator that influences such properties as dissolved oxygen levels, and can be modified by several factors related to forest harvesting, including increased solar radiation at the water surface. Sediment concentrations can impact the quality of drinking water and habitat quality for fish spawning. Sediments can also act as a vector for some nutrients and contaminants. Sedimentation is mostly a result of the construction of roads and stream crossings for logging operations. In general, sediment yields peak shortly after harvesting and decline with forest regrowth. The maximum sediment supply to streams, however, may not occur immediately after harvesting, particularly in the case of landslides.

Nutrient and contaminant levels in forested streams do not all react in the same way to forest harvesting, although there are general trends for most water chemistry indicators; and, in many cases, a lag is observed between forest harvesting and biogeochemical response. For example, loss of vegetation causes nutrient and contaminant concentrations to increase followed by a decline as result of uptake by forest regrowth.

Canada's national hydrological datascape

Knowledge generated by research regarding hydrological processes of forests and the effects of forest management activities on these processes are fundamentally based on hydrological datasets. These datasets provide forest hydrologists and managers with spatially distributed information about water quantity and quality, which can be used as a baseline to measure the response of hydrological systems to forestry operations within a watershed. The majority of Canada's forested watersheds are not gauged; thus managers must make informed decisions with the data that are available. Currently, there are two main sources of hydrological information in Canada:

- 1. Experimental study sites targeting locally-focused hydrological issues and/or research; and
- 2. Water Survey of Canada (WSC) river and lake hydrometric gauging sites (for discharge and lake level monitoring).

Experimental study sites

Many of the Canadian experimental watersheds were established by Federal and Provincial agencies along with university scientists in the 1970s. Several hundred scientific papers have resulted from these research watersheds at the local scale, but there have been few attempts to place these results in a national context and extrapolate results from these studies to other sites that are not gauged. One of the most valuable research efforts is the HydroEcological Landscapes and Processes (HELP) project. It identified 46 research sites across Canada and compiled, organized, and synthesized the knowledge into a framework to quantify hydrological, geomorphic, and ecologic processes in forests. Most of these 46 sites measure streamflow at the watershed outlet water temperature, lake out-flows, as well as air temperature and precipitation on a daily basis. Approximately half of the HELP's research sites are still active, many with invaluable long-term data records; however, maintaining funding for these sites has been difficult.

Water Survey of Canada (WSC)

The national Water Survey of Canada hydrometric archive operated by Environment Canada contains more than 2000 stations in forested regions across the country. Approximately 700 stations are still active, although there are gaps and areas of uncertainty in data records. Small basins in the boreal forest have only minimal coverage with few sites having records greater than 35 years.

To move forward, it is important to consolidate existing data and to increase data availability. The number of active gauging stations and experimental watersheds are decreasing, but developing relationships among government, university, and local stakeholder groups can help maintain or even improve upon the resources currently available.

Methods used for hydrological research

The earliest forest hydrology studies used field based methods to study hydrological processes and how they are affected by management activities; however, the recent proliferation of powerful desktop computing coupled with new remote sensing instruments has opened up a completely different vantage point for understanding and managing forests.

The fields of digital terrain analysis, remote sensing and hydrological modelling have great relevance for understanding the hydrological patterns and processes of forested ecosystems. These tools can be readily integrated into forest planning strategies since they are applicable to the study and management of forests at broad spatial and

long temporal scales. This does not mean, however, that the classic monitoring framework is obsolete, since digital tools will always rely on field corroboration. The future of both forest hydrology and forest management, from a hydrological perspective, will require an integrated planning and monitoring framework using all these tools, at a range of spatial scales from hillslopes to continental drainage basins.

Bird's-eye-view of forest hydrology: Novel approaches using remote sensing techniques

Remote sensing uses devices that detect electromagnetic radiation to observe phenomena from a distance. A variety of remote sensing techniques can be used to estimate components of the water budget at different spatial and temporal scales, giving forest managers additional information regarding the distribution of water in forest catchments to facilitate planning decisions.

Most early remote sensing sensors were optical; however, radiation in those wavelengths cannot penetrate clouds and vegetation. Consequently, these sensors are primarily used to monitor open areas of water such as lakes and wetlands. Alternatively, microwave radiation can penetrate vegetation and clouds, allowing a more un-restricted observation of forested landscapes. Both of these sensors measure the reflectance or backscatter of electromagnetic radiation from the objects that they observe. Selecting the appropriate sensor is an important first step in any remote sensing study. Some of the factors that managers should consider are the record-length, spatial resolution of imagery, and how radiation will interact with the object of interest and the atmosphere. Unfortunately, most airborne and satellite sensors were not specifically designed for hydrological applications, so data must be analyzed with care. Often, multiple sources of imagery can be combined to answer specific questions.

Visible and infrared imagery, and more recently microwave sensors can provide some information about how water enters a forested catchment through rainfall; the spatial resolution, however, is usually quite coarse. Visible and infrared imagery is also useful to determine the leaf area index which provides a good estimate of the amount of rainfall interception by the canopy. This is one of the most useful applications of remote sensing in hydrology and is used extensively in hydro-ecological models. Light Detection And Ranging LiDAR data can also provide information about the canopy structure, including gaps, height, soil bulk density, and surface area.

Remote sensing is able to provide new information about water stored in a landscape. While optical sensors can be used to estimate the area of surface water in a landscape, active microwave sensors and imaging radars can penetrate canopies and clouds and operate during day or night, giving more and higher resolution information about wet areas beneath the canopy. It is difficult, however, to monitor groundwater using remote sensing because the electromagnetic radiation used by current sensors cannot penetrate the ground. Alternative methods can be used to estimate groundwater recharge and discharge.

There are also a variety of methods to measure the different ways that water leaves forested catchments. Satellitederived vegetation indices, surface temperature and surface albedo have been used to map daily evapotranspiration at approximately 1-km spatial resolution, and much work is being done to improve these estimates. A disadvantage, however, is that discharge can only be estimated in rivers that are wide enough to be detected by sensors.

One particularly useful application of remote sensing for forest management is in the designation of hydrologically relevant buffer zones, which mitigate the effects of land use activities on nearby surface waters. Remote sensingderived maps can be used to assess the organization of surface flowpaths prior to the design and placement of such buffer strips.

To move forward, it is essential that new sensors, especially microwave, microgravity, and airborne geophysical sensors, are designed with the specific purpose of sensing hydrological phenomena. There is also a wealth of existing remote sensing data that in many cases is either inaccessible, discarded due to storage issues, or goes unused due to complexity of interpretation. A coordinated public and private effort is needed to archive these data and increase public access. There also needs to be increased interdisciplinary training for non-experts so that remote sensing analysis can be adopted by a broader audience.

Digital terrain analysis approaches for tracking hydrological and biogeochemical pathways and processes in forested landscapes

Digital terrain analysis (DTA) is an increasingly valuable tool used to map the movement of water and nutrients across forested landscapes. DTA holds great promise for forest managers since it can be used to identify critical features that should not be disturbed by management activities. Currently, DTA is being used to map hydrologically sensitive areas (e.g., wetlands and small streams) and minimize the impact of placing roads, culverts and cut blocks.

DTA is effective because in many landscapes, topography controls water flow by directing water from high elevations to low elevations due to gravity. Topography also forces water to converge and diverge due to the shape of the land surface. However, the dominance of topography as a primary control on water flow needs to be carefully considered for each landscape as there could be other biophysical factors that influence water flow, such as geology, soils, and vegetation.

In order to understand how DTA can help in the management of forests, it is important to have a basic understanding of the principles of this technique. In most cases, a square grid with a defined resolution is laid over top of an area, and for each cell the elevation is measured using photogrammetry, laser altimetry (LiDAR) or interferometric techniques. The resultant surface is called a Digital Elevation Model (DEM). DEMs are used to model how a drop of water would move across these digital surfaces. The preferred source of DEMs is LiDAR, as the laser beams can penetrate the forest canopy and record very accurate surface elevations. It is important to consider the spatial resolution of DEMs since most DTA techniques are very sensitive to the spatial resolution at which they were derived. The hydrological features being modeled should be properly matched against the spatial resolution of the DEM, since overly-coarse resolutions will misrepresent patterns, while overly-fine resolutions will contain too much detail to be useful.

DTA techniques have been developed to map hydrological pathways and areas of hydrological storage. For example, LiDAR-derived DEMs have significantly improved our ability to map canopy-shaded rivulets and wetlands within headwater forests. Not only do we have a better idea where streams begin, but we also have better assessments of where water is stored on the surface and how these different features are connected via surface saturated areas. This improved mapping of hydrological features has allowed for improved predictions of streamflow, including the different components of the hydrograph and water transit times through catchments.

DTA can also be used to identify hydrological controls on the formation of biogeochemical pools. Many of the DTA techniques that track the movement of water from land to surface waters are applicable for monitoring nutrients due to the close interrelationship between water and nutrient movement. For example, there is a rich literature on applying DTA techniques to predict the export of nutrients from catchments. DTA has also found use in the prediction of land-atmosphere transfer of nutrients with considerable focus on the production of greenhouse gases including carbon dioxide, methane, and nitrous oxide.

Looking into the future, the advances in the vertical and horizontal accuracy of DEMs must continue and be made broadly available across different forest types. Additionally, an improved integration and validation with field-based measurements, remote sensing, and distributed hydrological modelling, complemented by the establishment of global benchmark datasets, will further improve the applicability of DTA data. Improved imaging of subsurface features, such as bedrock topography, would substantially improve our ability to model hydrological processes in terrain where surface topography is not the dominant control on water flow (e.g., boreal plains).

Virtual ecosystems: Using hydrological models to understand the past and predict the future of water in forests

Quantitative ecological models are mathematical descriptions of ecosystems processes. These models serve as virtual laboratories, providing opportunities to manipulate components and functions more easily than in natural systems. In forest hydrology, quantitative ecological models can be used to understand the water and associated nutrient and sediment cycles.

In order to select the appropriate model when using quantitative methods, it is important to have clear what the overall goal of modelling is. It is also important to consider the context, function, performance, and evaluation of available models. Defining the model context places the proposed modelling exercise into a larger scientific, management, policy, and geographic context; the latter characterized by scale and an expectation of uncertainty. The function of the model determines which part of the hydrological cycle needs to be considered, what level of detail is needed to describe the hydrological processes, and what modelling resources are required. Once a potential model has been selected, its performance is optimized by adjusting key parameters to improve model fit. Finally, it is important to evaluate the model by inspecting and analyzing the output and determining the error of the computed data.

Hydrological models can be used in different ways to aid in the decision making process of managers:

- 1. Models can be used to define the reference condition of forests that have not been monitored for long enough for determining the range of natural variation in hydrological properties (usually streamflow). Hydrological models can also be used for virtual experiments such as simulating the possible impacts of harvesting activities. Experiments can be designed to separate the effects of multiple disturbances, such as climate, fire, harvesting, or acid rain.
- 2. In terms of forecasting, models can be used to simulate possible futures under changing environmental, management, and climatic scenarios. For example, global circulation models are used to predict future climate scenarios that will potentially impact the hydrological cycling of water within forests.

It can be difficult, however, to apply these models for planning and management at the regional and local scales. There are several modelling methods that allow continental climatic information to be downscaled to local levels, but it can be difficult to incorporate all of the effects of climate change accurately, including all the ecosystem processes that will be altered, into a locally applicable model.

Recommendations

Based on the current state of science and analytical tools reviewed in this document, we make the following recommendations in order to improve our future understanding of forest hydrology in Canada:

- Create a national data archive for forest hydrology: A centralized database would encourage cross-site comparisons and synthesis of meta-experiments. This will greatly expand knowledge generation and increase the incorporation of these resources into forest management planning and operations. Due to most hydrological data being funded by federal and provincial governments, we advocate for the public distribution and access of these dataset at no cost.
- **Continue existing and establish new monitoring networks**: Many crucial questions in forest hydrology, especially those related to climate change, can only be addressed by analyzing long-term datasets. Canada needs to take a lesson from the USA where there is a network of long-term observation sites.
- Encourage and adopt emerging technologies for quantitative analysis: The integration of remote sensing imagery in hydrological models will continue to be a critical research area. Further integration of digital terrain analysis, remote sensing, and distributed simulation modelling data sets using GIS-based spatial decision support systems will be especially important for managers and other decision makers. We need improved sensors for in-situ sampling as well as for airborne and satellite remote sensing of various hydrological phenomena.
- **Promote integration of datasets at relevant scales**: We need to integrate spatially explicit forest modelling processes with hydrological tools to enhance our ability to upscale results from headwater catchments to regional drainage basins, consider in-stream as well as lake and wetland processes, and forecast the cumulative effects of forest management activities on water resources through time.

- Develop a national watershed classification system: An ecohydrology-based watershed classification system would enable planners to recognize the spatial variability in and between different catchments to refine forest management prescriptions based on site conditions and associated hydrological processes, as well as to minimize the potential for adverse effects from harvest or renewal activities on water resources and lotic ecosystems.
- **Promote hydrological training and knowledge transfer**: The goal of this report was in part to promote the application and adoption of the hydrological knowledge and techniques presented herein within forest management. We need enhanced knowledge transfer from product developers and users in digital terrain analysis, remote sensing, and distributed simulation modelling to forest hydrologists and managers.

Scientific theory, data and techniques for conservation of water resources within a changing forested landscape irena creed et al. 2011 | state of knowledge report | sustainable forest management network

Introduction

The importance of conserving and protecting Canada's water resources cannot be overstated. Canada is well endowed with both forests and water, which provides the country with a social and economic advantage that will only grow during the 21st century. It is essential that all parties to water resource management, from local communities through industry and all levels of government, improve our collective stewardship of Canada's most valuable natural resource: water. Since forested areas provide Canadian communities, industries and ecosystems with the large majority of their water, sustainable forest management has a crucial role to play in conserving and protecting water resources.

Since forested areas provide Canadian communities, industries and ecosystems with the large majority of their water, sustainable forest management has a crucial role to play in conserving and protecting water resources.

The foundation of sustainable forest management (SFM) is the application of sound scientific principles applied through cutting-edge tools and techniques, all underwritten by high quality data. This applies to all aspects of forest management, from stand-level prescriptions to government policy. SFM endeavors to manage the forest resource to ensure that current forest values, goods and services are provided now and in the future. This requires both terrestrial and aquatic ecosystems to be considered in terms of ecosystem structure, function, and resiliency at multiple scales, both over time and across space.

The foundation of sustainable forest management (SFM) is the application of sound scientific principles applied through cutting-edge tools and techniques, all underwritten by high quality data.

This report focuses on the science behind sustainable management of forests from the perspective of conserving water resources and minimizing adverse effects due to forest management activities. After reviewing the state of knowledge of the effects of harvesting on forest hydrology and biogeochemistry, we provide an overview of current and emerging tools and technologies for managing water resources, along with a discussion of Canadian water research sites and data repositories.

The tools and technologies presented include remote sensing, digital terrain analysis, and watershed modelling techniques to address many of the hydrological issues faced by forest managers and policy makers today. They provide guidance for future investments in research and monitoring. This report will be useful to forest managers engaged in such activities as the development of strategic forest management plans, research and monitoring programs, and the development and reporting of indicators of SFM related to aquatic systems. Policy makers will also find it useful in terms of program development for provincial databases and inventory programs, as well as for the development of policy and guidelines pertaining to forest management planning and state of the forest reporting.

Hydrological principles: a framework for policy and practice

This document builds on a companion State of Knowledge (SOK) report on hydrological principles for conservation of water resources within a sustainable forest management framework entitled *"Hydrological principles for conservation of water resources within a changing forested landscape"*. A suite of hydrological principles is shown in Table 1; these principles provide a foundation for the development of sustainable forest management strategies to conserve water resources. The hydrological principles are intended to provide a framework for policy and operational practices designed to ensure that water resources are conserved within a sustainable forest management approach.

If forest managers develop SFM strategies and practices based on hydrological principles, they can expect their operations to be less risky in terms of environmental effects. This results in aquatic systems that are resilient to disturbance and environmental perturbations. Additional advantages include a communication framework to inform the public about water conservation strategies during consultation processes associated with forest management plan development, preservation of social license to operate, and continued market access. Similarly, the hydrological principles can provide a policy framework to ensure that aquatic elements of the landscape are sustainably managed, and promote resilient aquatic ecosystems.

If forest managers develop SFM strategies and practices based on hydrological principles, they can expect their operations to be less risky in terms of environmental effects.

Computer-based data and tools

Computer-based digital data and tools are a prime focus of this report because they lend themselves very well to planning and management applications at the scales at which forest management decisions are made. Although these digital data and tools do not replace the need to make ground-based measurements, they can be used for the characterization and prediction of hydrological patterns and processes at landscape scales. For example, the mapping of surface hydrological flowpaths using ground-based methods is prohibitively expensive and therefore unfeasible at the landscape scale; while the same task using digital terrain analysis and/or airborne or satellite remote sensing is becoming a routine process.

Of course, all of these digital data and tools require ground-based measurements either for their parameterization, calibration or corroboration; thus these varied data sources are completely complementary. It is the combined use of ground, airborne and satellite remote sensing and simulation datasets that results in tremendous insights in the hydrological sciences, as well as successful forest management strategies.

The combined use of ground, airborne and satellite remote sensing and simulation datasets results in tremendous insights in the hydrological sciences, as well as successful forest management strategies.

Geographical context

The data and tools presented in this report must be placed in a geographic context due to the highly variable nature of Canada's forests and associated hydrology, climate and landforms. In fact, acknowledgement of spatial and temporal factors that influence hydrological processes is one of several key hydrological principles for managing water resources (Table 1).

The influence of these geographic factors means there is no single data source or standard tool that can be used without modification for hydrological assessments. One must always be wary of temptations to apply

Hydrological Principles	Management Actions
1. Delineate hydrological system boundaries: Consider the entirety of the hydrological system within which management actions take place.	A) Delineate hydrological system boundaries based on knowledge of dominant hydrological flowpaths (many hydrological systems will coincide with topographic boundaries but in some places other factors control hydrological response units).
2. Conserve critical hydrological features: Minimize disturbance to hydrological features with critical source, transfer and storage functions.	A) Minimize disturbance to soils, especially within or near source areas that focus the recharge of water into subsurface pathways.
	B) Minimize disturbance in filter areas around streams, wetlands and lakes, and other sensitive sites (required buffer width will depend on dominant hydrological processes in given locale to maintain water quality of receiving water bodies).
	C) Minimize disturbance to storage areas (such as wetlands and ephemeral saturated areas).
3. Maintain hydrological connectivity: Minimize disruptions to water, sediment, nutrient flows within terrestrial system.	A) Consider the interconnectedness and interdependence of water pathways through watersheds when developing management plans (i.e., look beyond the forest stand and consider where the stand occurs with respect to the watershed and water flows).
	B) Locate roads, bridges, culverts and harvest areas to ensure surface and subsurface hydrological connectivity is maintained and flow is neither impeded nor enhanced.
4. Respect temporal variability: Acknowledge temporal (historic) factors that	A) Recognize there is natural variability in hydrological processes at multiple scales from daily to multi-decadal.
influence hydrological processes.	B) Recognize there is human induced variability in hydrological processes of different severity (from past management practices to climate change).
	C) Recognize the timing, frequency and magnitude of extreme events may be changing because of the interplay between natural and anthropogenic factors that are difficult to separate.
5. Respect spatial heterogeneity: Acknowledge spatial (geographic) factors that	 A) Consider how scale influences dominance of hydrological processes (moving from headwaters to regional basins).
influence hydrological processes.	B) Consider how geographic context including climate, bedrock geology, surficial geology, soil type and depth, and topography influences dominance of hydrological processes and patterns as well as forest type and age.
6. Maintain redundancy and diversity of hydrological form and function: Manage with the ethos that redundancy and	A) Consider watershed functions that might be most impacted by future extreme events and plan to protect features that perform those functions.
diversity of hydrological form and function contribute to a forest that can absorb outside	B) Consider multiple ecosystem services when assessing tradeoffs in making development choices.
disturbances.	C) Consider the interactive nature of the hydrological system with climatic, geomorphic, ecologic and socio-economic systems.

methods and tools developed or tested in one region to another region without local calibration. Similarly data extrapolation from areas with different climate, bedrock and/or surficial geology is likely not appropriate.

The influence of geographic factors on hydrological processes means there is no single data source or standard tool that can be used without modification for hydrological assessments.

Structure of the report

In this report, the reader is first introduced to the state of scientific knowledge regarding the effects of forest harvesting on hydrology and biogeochemistry. Subsequently, we provide a brief description of existing ground based hydrological datasets available in a national context. These datasets served in part to generate the current state of knowledge and which hopefully will serve us for many more years to improve our understanding of forest hydrological processes. The subsequent three chapters follow a sequence from digital data and tools that are amenable for the static characterization of landscapes to more dynamic characterization. Here are brief synopses of the chapters:

- Chapter 2 synthesizes research examining the hydrological effects of forest harvesting by assessing areas of agreement and disagreement, as well as limitations.
- Chapter 3 provides an overview of potential sources for hydrological data in Canada, and the locations and basic attributes of long-term research or monitoring sites where hydrological data have been collected. Gaps in the current hydrological data sources are highlighted along with possible ways to fill those gaps.
- Chapter 4 examines novel approaches for collecting hydrological data and their analysis using remote sensing techniques.

- Chapter 5 details recent developments in digital terrain analysis techniques for tracking hydrological and biogeochemical pathways and processes in forested landscapes. This section includes ways to link science to practice and methodologies for operationalizing digital terrain analysis for application to forest management planning.
- **Chapter 6** combines remote sensing techniques and digital terrain analysis with **watershed modelling** to look at watershed-level assessments. The data requirements and output of these models are examined in terms of forest management planning requirements.
- Chapter 7 contains conclusions and recommendations to assist resource managers to integrate hydrological assessments into their forest planning and practices. It also identifies research and monitoring priorities to guide future research endeavors.

BOX 1

Terminology: Catchment, watershed, and drainage basin

Hydrologists use the terms catchment, watershed, and drainage basin interchangeably. All three terms signify an area of land where water from rain and melting snow or ice drains down the contributing hillslopes into a body of water, such as a stream, river, lake, or ocean separated from adjacent systems by a drainage divide. The terms refer to both the land surfaces from which water drains and the streams, rivers and lakes that convey the water.

In this report we follow this custom, although we try to reserve the use of the word catchment for small, headwater watersheds and basin for large, regional watersheds.

The effects of forest harvesting on forest hydrology and biogeochemistry¹

Copyright permission to re-publish Chapter 2 was only provided for the printed version of this document. To access the full text, tables, and figures included in this chapter please refer to the original Springer publication:

http://www.springerlink.com/content/978-94-007-1362-8/contents/#section=906241

¹ Buttle, J.M. 2011. Chapter 33: The Effects of Forest Harvesting on Forest Hydrology and Biogeochemistry. In: Levia, D.F., Carlyle-Moses, D.E. and Tanaka, T. (Eds.), *Forest Hydrology and Biogeochemistry: Synthesis of Past Research and Future Directions* (pp.659-674). Ecological Studies Series, No. 216, Springer-Verlag, Heidelberg, Germany. **Re-published with permission of Springer-Verlag**.

22 Scientific theory, data and techniques for conservation of water resources within a changing forested landscape irena creed et al. 2011 | state of knowledge report | sustainable forest management network
3.0

Canada's national hydrological datascape

"Hydrology is a difficult science because it deals with open systems that are very difficult to observe. When we do not yet have an adequate measurement technique for the rainfall falling on a catchment, or for continuously determining the increments of discharge along a stream channel, or measuring the changes in bulk storage in the sub-surface; then doing hydrological science rigorously is going to be difficult."

BEVEN (2008)

3.1 Introduction

Forest managers require high quality datasets to aid and improve their decision making and minimize the effects of forest management activities on hydrological systems. In an ideal world, every manager would have access to spatially distributed water quantity and water quality datasets that would provide not only a baseline to measure the response of the hydrological system to forestry operations in a watershed, but also typical hydrological responses under best management practices.

Access to high quality datasets greatly facilitates adaptive management and improves response time needed to ameliorate negative impacts.

Such information would greatly facilitate adaptive management (e.g., continuous improvement or 'learning while doing' through the monitoring and testing of predicted forest management activity responses or forecasts). It would also improve response time needed to ameliorate negative impacts.

Changes to the hydrological system can be measured at different points along a flowpath. In uplands, soil moisture is a good integrative variable; in lowlands, where water is channelized, water can be measured as streamflow or lake level. Ideally, streamflow is measured at multiple points along a stream network and not just at the watershed outlet.

Unfortunately, most of Canada's forested watersheds are currently ungauged (i.e., streamflow measurements are not taken), which is not surprising given the vast area covered by forests.

Most of Canada's forested watersheds are currently ungauged, therefore, the best strategy is to use data collected at select watersheds across the country and extrapolate knowledge from gauged to ungauged basins. The limited information available makes it challenging for forest managers to make informed decisions regarding hydrological effects when designing management scenarios or predicting future water resources under a changing climate. In the absence of specific datasets, the best strategy is to use data collected at select watersheds across the country and extrapolate knowledge from gauged to ungauged basins.

In Canada, there are two main sources of hydrological information available for forested areas:

- 1. Experimental study sites targeting locally-focused hydrological issues; and
- 2. Water Survey of Canada (WSC) river and lake gauging sites (for discharge and lake level).

The aims of this chapter are to:

- review in detail these two main sources of hydrological data for forested areas,
- show the information gaps that still exist, and
- make recommendations regarding the future of hydrological data to improve the sustainable management of Canada's forests in terms of their water resources.

3.2 Experimental watersheds

Hydrological knowledge of Canada's forests has advanced tremendously thanks to research conducted at multiple experimental sites established across the country since the International Hydrological Decade of the 1970s. These sites were generally established by federal and provincial agencies, with frequent participation by university scientists, and most are relatively small (median size 3.5 km²).

Efforts have been made during the last 30 years to compile the datasets and draw conclusions from the results of these studies (Bosch and Hewlett 1982, Hetherington 1987, Buttle et al. 2000, 2005, 2009), but there have been few attempts to place these findings into a national context to facilitate extrapolation of results to other sites of interest located in ungauged stations. One exception is the HydroEcological Landscapes and Processes project (HELP) project, funded by the Sustainable Forest Management Network (SFMN), one of Canada's Network for Centres of Excellence. Hydrological knowledge of Canada's forests has advanced tremendously thanks to research conducted at multiple experimental sites; however, few attempts have been made to place these findings into a national context and to facilitate extrapolation of results to other ungauged sites.

The HELP project - synthesizing hydrological data from forest sites

The objective of the HELP project was to compile the disparate hydrological data from experimental studies across Canada, and organize and synthesize the know-ledge into a framework for quantifying hydrological, geomorphic and ecologic processes of forest landscapes (Krezek et al. 2008).

The objective of the HELP project was to compile the disparate hydrological data from experimental studies across Canada, organize and synthesize the knowledge into a framework for quantifying hydrological, geomorphic and ecologic processes of forest landscapes.

A central component of the HELP project is the development of a hydrological database that pulls together diverse isolated pieces of information to form a coherent picture of the water resources in Canada's forests. Criteria for inclusion of research sites include the following:

- 1. At least one hydrological indicator is measured;
- 2. There is published reference to the site;
- 3. Sites are primarily forested; and
- 4. Site location information, experimental documentation and study dates must be available.

Based on the above criteria, 46 research sites have been identified (Figure 4 and Table 3).



Figure 4. Location of forest hydrology experimental study sites in different ecozones of Canada (modified after Krezek et al. 2008).

These sites are distributed across the country and represent most of the major forested ecozones in Canada, except the more northerly forests and grass mixed forest types (Hudson Plains, Taiga Shield, Boreal Cordillera, and Taiga Plains) (Figure 4). Most sites have experienced some type of forest management activity, predominantly clearcut harvesting. Wildfires were the dominant disturbance in three sites (sites 9, 13, 17, all in western Canada). Twelve undisturbed sites (sites 6, 19, 21, 22, 31, 32, 34-36, 45) and two older second growth sites (sites 4, 11) are also included, serving the important role of reference sites. Individual metadata pages have been developed for 24 of the 46 research sites. These include a description of the landscape of each site, the experimental design, and contact information (Figure 5). They allow comparison of hydro-meteorological characteristics of research sites in different landscapes across the country, and indicate the status of field-based studies of forest hydrology.

Given the importance of these databases, we believe that renewed institutional support is to finish the compilation and archiving of records of these critical Canadian databases is required. (See http:// canforhydro.org for status of the HELP project).

Table 3. Basic descriptive information on forest hydrology experimental study sites organized by ecozones	Table 3.	Basic descripti	ve information on for	est hydrology expe	rimental study sites or	anized by ecozones
---	----------	-----------------	-----------------------	--------------------	-------------------------	--------------------

ID	STUDY SITE	Latitude (dd)	Longitude (dd)	Min Area (ha)	Max Area (ha)	No. Basins	Forest Status	Ecozone
1	Russell Creek, BC	50.18	126.16		3100	1	Logged	Pacific Maritime
2	Carnation Creek, BC	48.92	125.00	19.4	988	5	Logged	Pacific Maritime
3	Roberts Creek/ Flume Creek, BC	49.42	123.64	16	61	3	Logged	Pacific Maritime
4	Upper Gray Creek, BC	49.60	123.60	_	14950	1	2 nd growth	Pacific Maritime
5	Malcolm Knapp/ UBC Research Forest, BC	49.25	122.55	12	89	6	Logged	Pacific Maritime
6	Wolf Creek, YT	60.53	135.12	1450	19500	3	Natural	Boreal Cordillera
7	Stuart-Takla, BC	54.87	125.38	3700	5200	5	Logged	Montane Cordillera
8	Nation River, BC	55.20	124.23	230	410	3	Logged	Montane Cordillera
9	Fishtrap Creek, BC	51.12	120.20	_	13500	1	Fire	Montane Cordillera
10	Upper Penticton Creek, BC	49.62	119.42	370	500	3	Logged	Montane Cordillera
11	Redfish Creek, BC	49.62	117.05	1500	2600	4	2 nd growth	Montane Cordillera
12	Cotton Creek, BC	51.35	118.85	105	1736	9	Logged	Montane Cordillera
13	Southern Rockies, AB	49.37	114.40	358	1315	7	Fire and Logged	Montane Cordillera
14	Streeter Creek, AB	50.11	114.07	51	589	4	Logged	Montane Cordillera
15	Marmot Creek, AB	50.95	115.00	212	940	4	Logged	Montane Cordillera
16	Tri Creeks, AB	53.16	117.24	1480	2820	3	Logged	Boreal Plains
17	Swan Hills/ Forest Watershed and Riparian Disturbance (FORWARD) Project, AB	54.20	115.78	13100	24800	3	Fire and Logged	Boreal Plains
18	Spring Creek, AB	54.94	117.73	1530	2100	2	Logged	Boreal Plains
19	Utikuma Lake/ Hydrology Ecology and Disturbance (HEAD) Project, AB	55.90	115.03	5	500	24	Natural	Boreal Plains
20	Terrestrial and Riparian Organisms, Lakes and Streams (TROLS) Project, AB	54.77	111.99	189	5669	3	Logged	Boreal Plains
21	Beartrap Creek/ Prince Albert Model Forest, SK	53.86	104.62	_	11600	1	Natural	Boreal Plains
22	Duck Mountain, MB	51.36	101.26	10300	423000	15	Natural	Boreal Plains
23	Experimental Lakes Area (ELA), ON	49.66	93.74	36	1393	6	Logged	Boreal Shield
24	Coldwater Lakes, ON	49.12	92.15	70	194	3	Logged	Boreal Shield

continued on the next page

30 30

ID	STUDY SITE	Latitude (dd)	Longitude (dd)	Min Area (ha)	Max Area (ha)	No. Basins	Forest Status	Ecozone
25	Current/ Escape Lakes, ON	48.56	89.24	_	20	1	Logged	Boreal Shield
26	White River, ON	48.06	85.81	130	820	3	Logged	Boreal Shield
27	Turkey Lakes, ON	47.01	84.42	3	1150	13	Logged	Boreal Shield
28	Batchawana River, ON	47.00	84.52	15	23617	30	Logged	Boreal Shield
29	Esker Lakes, ON	49.38	81.01	n/a	20000	14	Logged	Boreal Shield
30	Little Abitibi, ON	49.07	81.03	29	9192	20	Logged	Boreal Shield
31	Dorset Experimental Lakes, ON	45.38	79.13	21	23	5	Natural	Boreal Shield
32	Hermine, QC	45.98	74.02	_	5	1	Natural	Boreal Shield
33	Haute-Maurice, QC	48.52	73.42	62	136	5	Logged	Boreal Shield
34	Lac de la Tirasse, QC	49.20	73.63	_	56	1	Natural	Boreal Shield
35	Lac Clair, QC	46.95	71.67	_	226	1	Natural	Boreal Shield
36	Lac Laflamme, QC	47.32	71.13	_	68	1	Natural	Boreal Shield
37	Ruisseau des Eaux-Volées Experimental Watershed (REVEW), QC	47.27	71.46	122	917	2	Logged	Boreal Shield
38	Charlevoix, QC	47.00	70.50	37	279	7	Logged	Boreal Shield
39	Cote-Nord, QC	50.72	67.50	83	409	7	Logged	Boreal Shield
40	Copper Lake, NL	48.83	57.83	114	124	4	Logged	Boreal Shield
41	Triton Brook, NL	48.60	54.59	_	23297	1	Logged	Boreal Shield
42	Nashwaak, NB	46.30	67.03	125	660	3	Logged	Atlantic Maritime
43	Catamaran Brook, NB	46.94	65.91	_	5200	1	Logged	Atlantic Maritime
44	Hayward Brook/ Holmes Brook, NB	45.87	65.19	136	1520	7	Logged	Atlantic Maritime
45	Kejimkujik National Park, NS	44.33	65.20	874	72300	4	Natural	Atlantic Maritime
46	Haliburton/ Gull River, ON	44.73	78.82	4.5	702	9	Logged	Mixedwood Plains



40

Canadian Forest Hydrology

canforhydro.org





Research Site Metadata

Great Lakes St. Lawrence Forest Region

Watershed Area(s): 3 to 1150 ha Mean Annual Runoff: 497 mm Mean Annual Precipitation: 1222 mm Mean Temperature: Jan -10.5°C; July 17.8°C





Climatic setting of the study period.



More Information Turkey Lakes Watershed Study Webpage Turkey Lakes Study Description

Figure 5. Example of metadata sheet compiled for Turkey Lakes Watershed experimental site by HydroEcological Landscapes and Processes (HELP) project (http://canforhydro.org).

Types of hydrological measurements made

There are some clear trends related to the hydrological measurements being made at these experimental research sites (Krezek et al. 2008). The primary hydrological variable collected at most of the study sites is streamflow; however, it is measured only at the outlet of watersheds. The most important hydrological variables for the remaining sites are water temperature and lake out-flows.

Most sites collect a minimum of air temperature and precipitation (rain and snow) data, unless a representative Meteorological Survey of Canada (MSC) site is located close by. Many research sites with primary meteorological stations also have additional meteorological measurements (e.g., wind, solar radiation, relative humidity) and hydrological data (e.g., snow studies, ground water, soil moisture). Twenty-two sites have aquatic ecology measurements (e.g., fisheries studies) and/or water quality measurements (e.g., water chemistry, dissolved oxygen).

Status of the sites and site records

In terms of activity, approximately half of the research sites are still active. Of the active sites, not all have ongoing scientific research, but they are being monitored for basic hydrological variables (Figure 6). Many of the sites have long-term data records, which are invaluable for studying ecohydrological change under variable climate conditions. Regrettably, maintaining funding for these sites is often difficult. In some cases, associations with other government agencies involved with data collection (i.e., MSC or WSC) have maintained the monitoring (e.g., Hayward Brook, Nashwaak, Kejimkujik) or preserved archived data records (e.g., ELA, Streeter Creek, and Marmot Creek), extending their longevity.

Some research studies, especially those that began in the late 1960s and early 1970s, have exceeded the length of most research professionals' careers, leading to many scattered or lost data sets. Therefore, a secondary benefit of the HELP project is the archiving of published data, making them readily available for future research applications.

Figure 6 indicates the duration of pre-disturbance, harvesting, and post-disturbance conditions

monitored by these studies. For example, the Turkey Lakes Watershed site has the longest pre-disturbance hydrological record of any site (16 years). Such long records are critical when assessing the natural variability in a region's climatology and hydrology. Data collected after forest disturbance can then be compared to the range of natural variability. This can help determine whether the hydrological system has been moved beyond the pre-disturbance range of conditions, and can be useful when setting targets in forest management planning and in scenario evaluation.

Long records are critical when assessing the natural variability in a region's climatology and hydrology. Data collected after forest disturbance can then be compared to the range of natural variability under pre-disturbance conditions.

3.3 Hydrometric networks: the Water Survey of Canada

The second major source of hydrological data for forest managers is the Water Survey of Canada's (WSC) national hydrometric network operated by Environment Canada (www.ec.gc.ca/rhc-wsc/). The network currently monitors streamflow (and chemical parameters at some stations) at 2,500 stations located across Canada and maintains an archive of 5,500 stations, of which approximately 700 are in forested regions.

The Water Survey of Canada's (WSC) national hydrometric network monitors 2,500 stations located across Canada and maintains an archive of 5,500 stations, of which approximately 700 are in forested regions.

41



Figure 6. Activity of watershed studies including (a) pre-disturbance and post-disturbance phases, and/or natural forests (thin lines) and (b) disturbance phases (thick lines); current as of 2007.

The median size of the WSC catchments being monitored is about 650 km², compared to a median of 3.5 km² for the experimental catchments (described in section 3.2), a two orders-of-magnitude difference. However, WSC data are collected for purposes other than assessing the impacts of forest management. Data from some of the WSC streamflow gauging stations have been used to assess the hydrological properties of forest landscapes and streamflow response to forest disturbance in parts of Canada, but it is very difficult to use streamflow records from larger basins to detect any hydrological consequences of forest disturbance (Buttle and Metcalfe 2000). In larger basins, regional factors, such as surficial geology and the presence of lakes and large wetlands, reduce the variability in streamflow metrics (Sanford et al. 2007) and obscure the impacts at the size of current forest harvest blocks (10-200 ha) (e.g., McRae et al. 2001; The Ontario Forest Management Guide for Natural Disturbance Emulation www.mnr.gov.on.ca/en/Business/Forests/ Publication/MNR_E000509P.html)^{2,3}.

It is very difficult to use streamflow records from larger basins to detect hydrological consequences of forest disturbance.

An analysis of the monitoring by WSC stream gauges in forested areas of Alberta, Quebec and New Brunswick (Buttle and Murray 2007) revealed that much of the boreal forest is not monitored at the small basin scale (<10 km²); and if monitored, the length of monitoring is short with very few sites having more than 35 years of record. There is also a lack of monitoring in the northern part of the boreal forest, even for larger basin sizes. This information is required in order to determine the relative impacts of forest disturbance and other factors (e.g. climate change) on the natural range of variability and frequency of extreme events in streamflow. Thus the data provided by the WSC's national hydrometric network is essential for monitoring and assessing water resources at broad scales, it is, with a few exceptions, insufficient for evaluating local forest management impacts.

This lack of small basin scale stream gauges is significant because it is very difficult to use streamflow records from larger basins to detect any hydrological consequences of forest disturbance (Buttle and Metcalfe 2000).

Much of the boreal forest is not monitored at the small basin scale and the length of monitoring is short with very few sites having more than 35 years of record. There is also a lack of monitoring in the northern part of the boreal forest, even for larger basin sizes.

3.4 Gaps in Canada's current sources of hydrological data

We need to be able to predict downstream effects of disturbance in headwater systems. We also need to be able to assess and predict the cumulative effects of resource development (forestry, oil and gas, mining) and other land uses on aquatic systems.

Unfortunately, the present streamflow monitoring network (experimental and WSC) does not provide adequate coverage of streamflow conditions to address many water resource research and management issues at the appropriate scales across Canada's various forest ecozones.

² McRae, DJ, Duchesne, L.C., Freedman, B, Lynham, T.J. and Woodley, S. 2001. Comparisons between wildfire and forest harvesting and their implications in forest management. Environ. Rev. 9: 223-260. Tables 1 and 2 compile data of cut block sizes for Canada. While there is considerable variation, all provinces, other than Ontario, limit cut blocks to less than 150 ha, sometimes much smaller (3.5 ha in PEI). In Ontario, cut blocks can exceed 260 ha, where justified, but the majority of cuts are less than 100 ha. That said, multiple cut blocks occur within a given watershed and can add up to significant fractions of a watershed.

³ The Ontario Forest Management Guide for Natural Disturbance Emulation (www.mnr.gov.on.ca/en/Business/Forests/Publication/MNR_E000509P.html) specifies that less than 80% in the Boreal and 90% in the GLSL must be less than 260 ha.

Even though experimental study sites collect a suite of hydrological measurements that are highly relevant to hydrological processes, most of the study sites are located in small (<10 km²) headwater catchments making the extrapolation of hydrological understanding to larger, regional catchments (>10 km²) problematic.

WSC gauging stations, in contrast, are located within much larger basins (>500 km²) and cover much longer time periods (Figure 7). While there are hydrological data at small and large basins, there is a dearth of gauging stations within basins in the 10-500 km² range.



Figure 7. Frequency distribution of watershed areas for experimental research sites in Canada compared with the WSC hydrological data base. The WSC sites have natural, unregulated measurements located within forested regions. Most of the experimental research sites (n=46) have more than one study catchment, which resulted in an overall sample size of 155.

There is a dearth of gauged basins in the 10-500 km² range. Also, for the gauged basins where forest harvest has occurred, the links between forest management activities and streamflow have not yet been clearly established.

There is a century of students from North America documenting the effects of forest management activities at small basin scales (cf. NRC 2008 and references within). However, scientists have not clearly established the link between forest management activities and streamflow at intermediate and large basin scales. This knowledge is important because it could lead to alternate practices to minimize effects at scales that are relevant to downstream water supplies.

Scaling relationships: from headwaters to regional basins

To understand the link between forest management activities and hydrological effects at local and larger scales, we need to establish scaling relationships between small headwater catchments and larger regional basins.

Questions such as: how does hydrological behaviour under 'natural' conditions propagate across scales? and how do harvesting related disturbances propagate? need to be answered. Insights related to these questions were provided in part by the SFMN-funded Scalable Indicators of Disturbance project. The project showed that in some of Ontario's forested landscapes the range of variability in streamflow properties (low flows in central Ontario [Sanford et al. 2007]; mean annual peak runoff in southern Ontario [Buttle and Eimers 2009]) may be considerable for small drainage basins, but begins to stabilize above a particular basin size threshold. Such behaviour is consistent with the Representative Elementary Area (REA) concept, which hypothesizes similarity in basin response above a threshold area.

The Representative Elementary Area (REA) concept hypothesizes similarity in basin response above a certain area. It also hypothesizes a tendency for larger basins to average the variability in local runoff patterns observed across smaller areas. This should be considered when attempting to predict effects based on data obtained at a different scale.

The REA concept also hypothesizes a tendency for larger basins to average the variability in local runoff patterns observed across smaller areas (Wood et al. 1988). This averaging behaviour needs to be considered when attempting to use the results from research basins to predict the cumulative effects of forest management activities for larger drainage basins.

The occurrence and magnitude of critical threshold areas should be examined for various forest landscapes. This could provide guidance to managers on the relationships between harvest disturbance levels and streamflow response.

3.5 Recommendations and conclusions

In order to effectively incorporate the hydrological principles into a forest management framework, forest planners and managers need access to relevant hydrological databases. Based on the current state of the hydrological datascape, we recommend the following steps be taken by the hydrological community as a whole.

3.5.1 Consolidate existing data and bring them under a common umbrella

A large volume of data relevant to sustainable forest management already exists, with the experimental studies conducted across the 46 headwater sites and the hundreds of WSC sites collecting streamflow data in Canada. However, many of the datasets collected at the experimental sites are being lost as people retire, studies are completed or discontinued, or as older storage devices become obsolete.

Many of the datasets collected at the experimental sites are being lost as people retire, studies are completed or discontinued, or as older storage devices become obsolete. These copilation efforts need to be continued.

The compilation effort that begun with the HELP project needs to be continued. All existing relevant datasets should be archived, along with the WSC and MSC datasets, which are already online. There are also other spatial datasets available (e.g., leaf area index, soil moisture, snow cover) which may inform managers about site conditions, hydrological flowpaths or soil moisture storage. Such datasets should be clearly identified and linked to the proposed hydrological portal as well.

3.5.2 Reverse the trend of closing ground based monitoring stations

Since the golden age of hydrometric monitoring during the 1970s, hundreds of stations have been shut down as cost saving measures. This trend clearly needs to be reversed since remote sensing of discharge is generally not possible.

More affordable monitoring, such as the installation of automated sensors, can be implemented to continue the data collection effort.

More affordable monitoring, such as the installation of automated sensors, can be implemented to continue the data collection effort.

In our opinion, government and university scientists need to work together with local stakeholders, including the public, to monitor their environment. For example, high schools could sponsor and operate sensors that measure hydrological variables to provide experience to students while decreasing overall monitoring costs. Data would be checked by professionals and compiled in centralized databases.

3.5.3 Complement ground based data with remote sensing data

Remote measurement of discharge is only possible on the largest rivers such as the Mackenzie or Athabasca; however, open water extent, saturated zone extent, and lake chlorophyll *a* can be measured remotely.

Other variables that can be useful in hydrological modelling, such as leaf area index and digital elevation models should also be linked to the hydrology portal for easy identification and overlaid with other spatial datasets. Remote sensing images provide an excellent source of data on characterizing present and past land cover in order to assist in teasing out the separate effects of climate variability and forest disturbance.

Remote sensing images provide an excellent source of data on characterizing present and past land cover in order to assist in teasing out the separate effects of climate variability and forest disturbance.

Landsat (a series of satellites operated by US government organizations used to gather optical information about the Earth's surface) databases go back to the early 1970s, providing the longest record of land cover at regional scales.

Information on land cover needs to be integrated with the hydrological datasets compiled for the experimental and WSC watersheds.

3.5.4 Integrate data with hydrological models to improve prediction in ungauged basins

In lieu of collecting new data, which is definitely needed in many parts of our forests, the existing databases can be used to build hydrological models of different complexity to improve prediction in ungauged basins. Model predictions can be constrained to give results with acceptable uncertainty bounds and can be combined with remote sensing data layers, such as soil moisture and snow cover, to provide forest managers with additional information to aid in decision making.

The existing databases can be used to build hydrological models of different complexity to improve prediction in ungauged basins. In conclusion, we advocate a strategic data collection regime that relies on multiple stakeholders and is interfaced with existing datasets. This will help build a comprehensive picture of our hydrological systems in forested landscapes across the country.

Bird's-eye view of forest hydrology: Novel approaches using remote sensing techniques⁴

Copyright permission to re-publish Chapter 4 was only provided for the printed version of this document. To access the full text, tables, and figures included in this chapter please refer to the original Springer publication:

http://www.springerlink.com/content/978-94-007-1362-8/contents/#section=906241

⁴ Sass, G. Z. and Creed, I. F. 2011. Chapter 3: Bird's-eye view of forest hydrology: Novel approaches using remote sensing techniques. In: Levia, D.F., Carlyle-Moses, D.E. and Tanaka, T. (Eds.), *Forest Hydrology and Biogeochemistry: Synthesis of Past Research and Future Directions* (pp.45-64). Ecological Studies Series, No. 216, Springer-Verlag, Heidelberg, Germany. **Re-published with permission of Springer-Verlag**.

49

Scientific theory, data and techniques for conservation of water resources within a changing forested landscapeIRENA CREED ET AL. 2011IRENA CREED ET

scientific theory, data and techniques for conservation of water resources within a changing forested landscape irena creed et al. 2011 | state of knowledge report | sustainable forest management network

 Scientific theory, data and techniques for conservation of water resources within a changing forested landscape

 IRENA CREED ET AL. 2011
 State of knowledge report
 Sustainable forest management network

scientific theory, data and techniques for conservation of water resources within a changing forested landscape irena creed et al. 2011 | state of knowledge report | sustainable forest management network

5 Digital terrain analysis approaches for tracking hydrological and biogeochemical pathways and processes in forested landscapes⁵

Copyright permission to re-publish Chapter 5 was only provided for the printed version of this document. To access the full text, tables, and figures included in this chapter please refer to the original Springer publication:

http://www.springerlink.com/content/978-94-007-1362-8/contents/#section=906241

⁵ Creed, I. F. and Sass, G. Z. 2011. Chapter 4: Digital terrain analysis approaches for tracking hydrological and biogeochemical pathways and processes in forested landscapes. In: Levia, D.F., Carlyle-Moses, D.E. and Tanaka, T. (Eds.), *Forest Hydrology and Biogeochemistry: Synthesis of Past Research and Future Directions* (pp.69-95). Ecological Studies Series, No. 216, Springer-Verlag, Heidelberg, Germany. Re-published with permission of Springer-Verlag.

Scientific theory, data and techniques for conservation of water resources within a changing forested landscape irena creed et al. 2011 | state of knowledge report | sustainable forest management network
76 scientific theory, data and techniques for conservation of water resources within a changing forested landscape irena creed et al. 2011 | state of knowledge report | sustainable forest management network

 $\begin{bmatrix} 0 \\ 0 \end{bmatrix} \\ \begin{array}{c} \text{Scientific theory, data and techniques for conservation of water resources within a changing forested landscape} \\ \text{Irena creed et al. 2011} & \text{State of knowledge report} & \text{Sustainable forest management network} \\ \end{array}$

Virtual ecosystems: Using hydrological models to understand the past and predict the future of water in forests

"At the present stage of hydrological science, hydrologic modelling is most credible when it does not pretend to be too sophisticated and all inclusive, and remains confined to those simple situations whose physics is relatively well understood and for which the modeller has developed a good 'common sense' within his primary discipline."

KLEMEŠ (1986A)

6.1 Introduction

Models are mental constructs of how the world functions. By definition they are approximations and simplifications of reality, capturing the essence of the way systems work. Models may range from qualitative/ perceptual models (with only pictorial or verbal descriptions of system components and functions) to quantitative/numeric models (with physics-based mathematical descriptions of system components and functions). In this chapter we refer to models specifically as quantitative descriptions of how ecosystems cycle water, nutrients and sediments.

Models can be viewed as virtual ecosystem laboratories, in which replicating and manipulating system components is infinitely easier than doing the same with real ecosystems.

The revolution in desktop computing has led to the proliferation of quantitative models that are used to understand the internal workings of ecosystems, but may also be used as predictive tools to test future "what-if" scenarios. In a sense, models can be viewed as virtual ecosystem laboratories, in which replicating and manipulating system components is infinitely easier than doing the same with real ecosystems. Models provide better control in understanding individual processes, and provide the ability to hindcast and forecast the behaviour of systems. The possibilities are limitless, but modellers should remember that despite their best efforts, models are just educated guesses about how the world works.

In forest hydrology, models can be used to understand the water cycle and associated nutrient and sediment cycles. They also have a large potential for informing the placement of access roads, bridges and culverts, harvest blocks, landings and buffers in forest planning.

Models can be used to understand the water cycle and associated nutrient and sediment cycles. They also can inform placement of roads, bridges and culverts, harvest blocks, landings and buffers.

Early work in models evolved from engineering requirements to properly design structures to withstand large floods. By the 1960s, research scientists had also realized the power of models for understanding catchment behaviour (Hewlett and Hibbert 1967). The driving questions of the day, then as now, concerned runoff processes (i.e., how much of the rainfall is turned into runoff and what is carried with it). The earliest catchment model was the Stanford Watershed Model (now called HSPF) (Crawford and Linsley 1966), a lumped parameter hydrological model able to simulate the primary hydrological processes. Since then, modelling has come a long way due to faster computing, more realistic representation of patterns and processes, and improved characterization of the earth's surface and subsurface.

The evolution of hydrological models is succinctly captured by Singh and Woolhiser (2002), who reviewed a comprehensive list of 69 hydrological models developed since the 1960s. The most sophisticated models can now address soil-vegetation-atmosphere transfers of water and energy, soil moisture storage, different runoff generation processes, shallow groundwater movement, and in-stream or in-lake processes. Hydrological models are increasingly being used to predict hydrological and biogeochemical responses to anthropogenic changes of system components (e.g., removal of vegetation during forest harvesting) (Tague and Band 2001b).

Models can assess the cumulative effects of anthropogenic activities in forests on the hydrological system.

A key strength of models from a forest management perspective is the ability to assess the cumulative effects on the hydrological system of anthropogenic activities in forests such as harvesting, building of roads and other linear features, as well as simulating the changes in forest composition due to climate driven forest dieback.

Despite the proven usefulness of models in tackling hydro-ecological problems, their utility in capturing the complex nature of ecosystem processes has been questioned, especially when used to predict system

90

behaviour outside of their calibration period and place (Beven 1993). There are four main criticisms with respect to ecosystem models in general:

- Most of the processes represented in models are derived at a point scale but applied to much larger scales;
- 2. There is a lack of data to parameterize models (especially spatially distributed models);
- 3. Most spatially distributed models need experts to run them; and
- 4. Models can sometimes give the same answer using entirely different parameter sets, leading many to question the veracity of the entire model structure.

The key to successful modelling is the selection of a parsimonious model driven by the question at hand.

The response from modellers is that in spite of the general shortcomings of models, there are models for many hydro-ecological questions that are useful for understanding and predicting ecosystem behaviour across a wide range of scales and geographies. The key to successful modelling is the selection of a parsimonious model driven by the question at hand: where simpler models exist they should be used (Rosbjerg and Madsen 2005). For example, if only water yield is needed, then a one-dimensional bucket model that characterizes discharge at the catchment outlet might be sufficient. On the other hand, more complex, spatially explicit models would be needed if the objective of the modelling exercise was to predict runoff post-harvest since the impact of harvesting depends on the location of the harvest block with respect to hydrological flowpaths.

Model evolution is an ongoing process; models are continually being improved.

Model evolution is an ongoing process. While there might be inherent limits to models in general, the continual improvement of models based on the

comparison of model predictions and observations increases the reliability of this ecosystem tool.

The goal of this chapter is to provide a synthesis on modelling for forest managers and hydrologists who are interested in exploring different hypotheses about system behaviour in forested ecosystems, as well as testing the possible outcomes of different management scenarios on water resources. Some of the key questions being tackled by modellers are related to water quality, including nutrient and sediment export from catchments, and water quantity, including peak flows, base flows and annual water yield within natural and managed systems. An overriding issue for scientists and managers alike is predicting the likely effects of climate change on the different components of the water cycle, from local to global scales.

Modellers are tacking questions related to water quality and water quantity. An overriding issue is prediction of the likely effects of climate change on different components of the water cycle.

This synthesis will focus on forested ecosystems in Canada, with an emphasis on how hydrological modelling is being used by forest hydrologists in science, management and policy. It comes on the heels of an excellent review by Beckers et al. (2009) who focus on hydrological models relevant for forest management and climate change applications in British Columbia and Alberta. We offer a broader scope geographically and application-wise, and focus on the potential of models in general rather than specific models.

Models are a key component of the digital hydrological toolkit, along with GIS, digital terrain analysis and remote sensing techniques. These tools can help society to manage multiple ecosystem services and aid in the drive towards long-term sustainability. We explore the themes of how models advance our knowledge of the natural variation of hydrological patterns and processes in forests, the influence of anthropogenic effects on this natural variation and the cumulative effects of climate change. We first review steps involved in the selection and application of hydrological models, then address model applications.

Models are a key component of the digital hydrological toolkit, along with GIS, digital terrain analysis and remote sensing techniques.

6.2 How to pick your model: a primer

Finding the appropriate model for a given purpose can be a great challenge, especially in today's model-saturated world where model abbreviations such as HSPF and TOPMODEL do not reveal much about their purpose or ease of use. While a standard model for forest hydrology would be nice, currently no model can be certified or "validated" to be applicable in any given catchment. As a result it is imperative to work through the exercise of picking the optimal model for each new modelling application.

Jakeman et al. (2006) offer a useful step-by-step modelling framework for the development and evaluation of environmental models. We have modified this framework and tailored it for forest managers and hydrologists who do not have the resources to develop or modify an existing model (Figure 17). There are a variety of considerations that have to be made to make an informed decision on model selection: context, function, performance, and evaluation. We start with the assumption that the hydrologist entering the process of selecting a model has a well-defined modelling goal.

6.2.1 Model context

Model goal and model context go hand in hand, since the goal of a modelling exercise is defined by its context. For example, the goal could be modelling peak flows given a changing climate, but the geographic location (e.g., northern Ontario), scale (e.g., first-order basin) and acceptable level of uncertainty (e.g., $\pm 15\%$ of water balance components) need to be specified in order for model selection to proceed. Defining the



Figure 17. Model evaluation framework for forest hydrologists.

model context places the proposed modelling exercise into its larger scientific, management and policy environment as well as into its geographic context characterized by scale and an expectation of uncertainty.

Modelling studies are driven by different goals. These range from basic hypotheses testing about how ecosystems work to more applied goals such as evaluating the effects of different management scenarios. Especially in applied contexts, more than one stakeholder may be involved with the modelling, and their individual expectations will need to be balanced with those of others.

Model goal and model context go hand in hand, since the goal of a modelling exercise is defined by its context.

The initial scoping for the modelling exercise should keep the issues of scale and uncertainty at the forefront. The question of scale will determine the geographic extent of the system being studied as well as the larger context it fits within. Scale, through its strong association with surface heterogeneity, also has a strong influence on model structure as discussed in the next section. As scale increases, processes that are important at finer scales can be omitted and model structures can be simplified. In applied contexts, such as evaluating the effects of different management scenarios, more than one stakeholder should be involved with the modelling, and their individual expectations will need to be balanced with those of others.

Acceptable levels of uncertainty have to be determined at the outset of the modelling study. For instance, models that serve to protect life and property (e.g., flood protection structures) require much lower uncertainty associated with the output than models that are used for understanding the natural behaviour of systems.

Scale and acceptable levels of uncertainty are important considerations to take into account while building a model.

6.2.2 Model function

Once the modelling context is defined, the stage is set for selecting a model that can answer the question under consideration at the appropriate scale with the appropriate level of uncertainty. Model selection enters the stage of considering model functionality, which asks the following critical questions:

- 1. Which part of the hydrological cycle needs to be incorporated in the model?
- 2. At what level of detail should the hydrological processes be described?; and
- 3. What will be the resource requirements for a model with the selected functional attributes?

Once the modelling context is defined, the stage is set for selecting a model that can answer the question under consideration at the appropriate scale with the appropriate level of uncertainty.

6.2.2.1 Model structure

Model structure is a determinant of model functionality. It is a multifaceted concept that considers process representation of the relevant parts of the hydrological cycle, the level of spatial and temporal detail (or discretization) needed to represent the processes, and the computational techniques needed to run the model (Table 9). Process representation and level of detail are in turn influenced by the climate, geology and topography of an area.

Process representation

Process representation describes the way sub-components of the system interact with other sub-components in terms of transferring energy and matter between them. For example, in forested ecosystems the canopy is an important sub-component as it intercepts precipitation (matter) and converts sunlight (energy) to be processed and used by other sub-systems. Models represent processes in three major ways:

- Empirically, using experimentally derived relationships such as linear regressions (e.g., US Soil Conservation Service (SCS) curve number for calculating runoff);
- 2. Analytically, using simplifying assumptions to derive solutions of the governing equations describing conservation of mass, momentum and/or energy (e.g., Green-Ampt equation for describing infiltration); and
- 3. Physically, using equations describing conservation of mass, momentum, and/or energy (e.g., Penman-Monteith for ET).

Process representation describes the way sub-components of the system interact with other sub-components in terms of transferring energy and matter between them.

Some models use a combination of approaches to represent different processes. There are special, geographically unique watershed processes that need to be considered as well. These include fog drip, rainon-snow events, lake and wetland storage and release. Some of these processes involve phase changes of water, and require careful consideration because of the associated transfers of matter and energy.

Level of detail for process descriptions

The level of detail for process descriptions depends on the type of geography as well as the required precision for model output (both in terms of space and in time). For example, drier climates will need to consider multiple storage compartments and their interactions.

The level of detail for process descriptions depends on the type of geography as well as the required precision for model output.
 Table 9. Structural properties of models characterized by spatial and temporal discretization, process representation

 (defined by relationships between system components and the outside world), and computational approach

Structural property of model		Different approaches to representing structural property			
Spatial discretization (horizontal)		Lumped Semi-distributed (hydrological response unit, elevation bands, hillslopes, sub-basins, topographic index, etc.)			
					Distributed (grid cells, patches based on intersection of differen GIS layers)
		Spatial discretization (vertical)	Soil layers	Not represented	
	Conceptual representation (unsaturated/saturated zones)				
	Explicit soil depths				
	Finite difference or finite element discretization (subsurface grid cells)				
	Vegetation layers	Not represented			
		Single layer (canopy)			
		Two layers (canopy, shrub layer)			
		Two or more layers			
Temporal discretization		Annual			
		Monthly			
		Daily			
		Hourly			
Watershed processes		Empirical approaches			
		Analytical approaches			
		Physical approaches			
Computational approach		Deterministic			
		Stochastic			

It is not necessary to enumerate all hydrological processes thought to occur in an area; instead the challenge is to find the dominant ones given the physical environment. More elaborate physical descriptions lead to higher degrees of complexity and usually greater demand for resources (both in terms of data and computational power), expert knowledge and finances. A wisdom passed down through the ages is encapsulated in Occam's razor, which says that if something can be described in simpler terms it is probably the more realistic explanation. The idea of parsimony is critical, but managers need to balance carefully the necessary process descriptions in order to get the right answers for the right reasons.

Discretization

Process representation is closely linked to model discretization, which describes how the model treats the spatial and temporal dimensions of the system being modelled (Table 9). The spatial dimension has both horizontal and vertical components. Horizontally, space may be:

- 1. Lumped (i.e., one or just a few units);
- 2. Semi-distributed (i.e., parts of the landscape with similar physical characteristics behaving identically); or
- 3. Gully distributed (multiple units each behaving under their own regime).

Process representation is closely linked to model discretization, which describes how the model treats the spatial and temporal dimensions of the system being modeled.

Distributed models are divided into parts based on a grid pattern or based on patches defined by uniqueness in geology, soils, climate and land use/land cover. In the vertical dimension, models may represent vegetation (e.g., canopy, trunk, shrub, and ground cover), soil and multiple surficial geology layers. In the temporal dimension, the most common timescales are daily, monthly and annual; however, some models focus only on individual precipitation events or on steady-state conditions.

The level of spatial and temporal discretization depends in part on the modelling question being asked. Unfortunately, the level of discretization is often driven by availability of data and not by science-based considerations.

Computational approaches

Different discretization and process descriptions might entail different computational implementations. Computationally, models can be categorized as deterministic or stochastic. With deterministic models, all model runs will give the identical answer given the same input data. In contrast, stochastic models will have different output for every model iteration due to the element of chance associated with the random selection of parameters from predefined distributions. Stochastic models are by nature preferable when testing the sensitivity of a model's performance to parameter distributions but do require long computation times.

Different discretization and process descriptions might entail different computational implementations. Computationally, models can be categorized as deterministic or stochastic.

Models and forest management

While most hydrological models can be used to answer fundamental hydrological questions regarding the water balance, only a small number of models are directly relevant to forest management. The most important issues for managers are planning and operations. Management has historically focused on maximizing biomass production with operational efficiency and minimal environmental effects (e.g., minimizing machinery downtime due to muddy conditions). However, managing for ecosystem services like water quantity and water quality is increasing, due to updated provincial regulations, forest certification requirements and the use of a criteria and indicators approach to SFM (Canadian Council of Forest Ministers 1995, Blumenfeld et al. 2009).

Addressing these management issues usually requires an explicit treatment of space since the position of harvest blocks, roads, and other disturbances determine their impact on the hydrological system. Embedded in the consideration of space is a requirement for high spatial resolution of input data when looking at the impact of linear features such as roads on hydrological flowpaths.

Different discretization and process descriptions might entail different computational implementations.

6.2.2.2 Model resource requirements

Models require data, computers and finances. The resources required are a direct function of model structure and complexity (Table 10). Data requirements usually create the most severe bottlenecks because data are scarce. On a practical level, data availability consequently often drives model selection given that most users will not have the ability or resources to collect more data in order to satisfy the requirements of perhaps more appropriate models (Beven 2004).

Data requirements can be broken into three categories:

- input data to force the model forward in time,
- spatial characterization data to describe the spatial differences between the different modelling units, and
- parameters with which to calibrate model equations.

	Low complexity	Medium complexity	High complexity	
Data requirements				
Climate forcing variables	Monthly precipitation and temperature	Daily precipitation and temperature	Hourly to daily precipitation and temperature	
Additional forcing variables	No additional meteorological forcings required	Additional meteorological forcings may be required	Additional meteorological forcings may be required	
Spatial data	No need for spatial data	Requires spatial data (DEM, soils, and forest cover)	Requires spatial data (DEM, soils, and forest cover)	
# of parameters	Less than 25 parameters	About 25 to 75 input parameters	Typically over 75 input parameters	
Calibration	Parameters are experimentally based (no calibration required); models suitable for ungauged basins	Minimal number of calibration parameters; some models applicable to ungauged basins	Medium to high number of calibration parameters; models applicable to gauged basins only	
Resource requirements				
Preprocessing	Low data preprocessing effort	Medium data preprocessing effort	High data preprocessing effort	
GIS analysis	Does not require GIS analysis	GIS analysis required for some models	GIS analysis required for most models	
# of people required	Can be completed by one person	Can mostly be completed by one person	May require project team	
Time requirements	Less than 2 weeks	About 2 weeks to 2 months	About 2 to 6 months	

 Table 10.
 Model complexity and its relation to data, resource and time requirements for any modeling exercise

Models require data, computers and finances. Data availability often drives model selection.

Climate data drive most hydrological models, and often consist of only precipitation and temperature time-series. In river models, runoff time series may suffice as input and many steady-state groundwater models are driven by constant head or flux boundaries.

Spatially distributed data describe the land surface, including topography, geology, and vegetation. There are multiple sources, but there is much weaker coverage in remote areas and of the subsurface. The real distinction between spatially distributed data and data that are used as parameters is that parameter values may end up being modified during the calibration process whereas the other descriptive datasets will not. Parameter values such as hydraulic conductivity come either from field measurements (i.e. from previous studies), or become adjustable model values (i.e. modified during the model calibration process).

Climate data drive most hydrological models. Spatially distributed data describe the land surface, including topography, geology and vegetation.

Data uncertainty is an important consideration in selecting input data; the old axiom "garbage in, garbage out" is very true. Model outputs will be as good as model input (Whitaker et al. 2003). At this point in the modelling process, there may be tradeoffs between a feasible model and the ideal model. The availability of the data will dictate some choices and may even change the overall purpose of the modelling exercise.

Data uncertainty is an important consideration; the old axiom "garbage in, garbage out" is very true.

Despite many technological advances, computational resource requirements are still an important consideration. Spatially explicit models covering large areas require extensive processing time (Table 10). Also, learning to assemble datasets and parameterize models can take some time. All of these factors play into modelling costs, which also can influence the type of model used.

Hydrological models for use in forest management

So what are the best models from a forest management point of view? In a recent review, Beckers et al. (2009) concluded there is no single "best" model for use in an operational forest management context. They identified nine models that could potentially be used for addressing forest management and climate change questions in British Columbia and Alberta (Table 11). These include one low-complexity model (WRENSS), four medium-complexity models (UBCWM, BROOK90, ForWaDy, UBC-UF), and four medium-complexity models (DHSVM, RHESSys, WaSiM-ETH, CRHM).

There is no single "best" model for use in an operational forest management context. Many cannot yet address questions of interest to forest managers, or are difficult and costly to apply. What is striking from Table 11 is that the components that most interest forest managers are the least developed and represent an important knowledge gap. For example, only the most complex models, DHSVM and RHESSys, have the capability to address the effect of roads on hydrological systems. On the other hand, distributed models such as DHSVM and RHESSys are difficult to apply and populate with realistic data, requiring significant time, expertise and cost. As a result they are only applied to a few sites by a few experts.

6.2.3 Model performance

Calibration

Once a model has been chosen and the necessary data compiled, the model is ready for calibration. Calibration involves adjusting (tweaking in modellers' terms) a few key parameters that have a strong control on model output. In most modeling exercises this results in only two or three sensitive parameters (Jakeman and Hornberger 1993). For example, in RHESSys, hydraulic conductivity and the decay of hydraulic conductivity with depth are the two parameters that are "tweaked" during calibration.

Once a model has been chosen and the necessary data compiled, model is ready for calibration. Calibration involves adjusting a few key parameters that have a strong control on model output.

While many of these parameters start out as measurable values in the field, their final values used in the model can be totally dissociated from reality, and are often called effective parameters. Some people have argued that these parameters capture an emergent property of catchments such as a catchment integrated hydraulic conductivity value (C. Tague, personal communication).

Assessing performance

After each calibration run, model performance is assessed based on visual inspection of the hydrograph and statistical assessment of goodness of fit comparing

Category	Purpose	Abbreviated model name	Model complexity	Watershed discretization	Simulation of forest harvesting	Forest growth	Road construction and management
Forest hydrology models	Annual water yield	WRENSS (WinWrnsHyd/ECA-AB)	Low	Lumped	Mixed	No	No
	Peak flow	UBC-UF Peak Flow Model	Medium	Distributed	Empirical	No	As Hortonian overland areas
	Water balance	BROOK90	Medium	Lumped	Mixed	No	No
		ForHyM	Medium	Lumped	Empirical	No	No
		ForWaDy	Medium	Lumped	Mixed	No	No
	Watershed hydrology	DHSVM	Medium	Distributed	Physical	No	Road drainage network and structures
		RHESSys	High	Distributed	Analytical	Yes	Road drainage network and structures
Other models	Water balance	HELP	High	Lumped	Mixed	Yes	No
	Watershed hydrology	ACRU	Medium	Semi-distributed	Mixed	No	No
		CRHM	High	Semi-distributed	Physical	No	No
		HBV-EC	High	Semi-distributed	Empirical	No	No
		HEC-HMS	Medium	Semi-distributed	Empirical	No	No
		HSPF	High	Lumped	Mixed	No	No
		PREVAH	High	Semi-distributed	Analytical	No	No
		PRMS/MMS	High	Semi-distributed	Mixed	Yes	No
		SWAT	High	Distributed	Mixed	Yes	No
		UBCWM	Medium	Semi-distributed	Mixed	No	No
		WaSiM-ETH	High	Distributed	Analytical	No	No
		Watflood	High	Semi-distributed	Mixed	No	No
	Groundwater -surface water models	InHM (HydroGeoSphere)	High	Distributed	N/A	No	1-D overland flow segments
		Mike-SHE	High	Distributed	Mixed	No	No
		MODHMS	High	Distributed	Empirical	No	No
	Soil erosion	WEPP	High	Semi-distributed	Mixed	Yes	No
	River basin models	SSARR	High	Semi-distributed	Empirical	No	No
		VIC	High	Lumped	Physical	No	No

Table 11. Models that have been useful in the study of forest hydrology (modified from Beckers et al. 2009)

model output (daily discharge in most instances) with observations. The most common goodness of fit metric is the Nash-Sutcliffe efficiency (Nash and Sutcliffe 1970), despite some investigators questioning its universal application (McCuen et al. 2006, Jain and Sudheer 2008).

After each calibration run, model performance is assessed based on visual inspection of the hydrograph and statistical assessment of goodness of fit comparing model output with observations.

While Nash-Sutcliffe and other goodness of fit metrics, such as the coefficient of determination and mean square error, may be low hurdles to test agreement between measured and simulated values (Sivapalan 2009), goodness of fit metrics and metrics of bias, together with good judgment, can help find the optimal parameters. Selecting the optimal parameter values is often done on a trial and error basis. Recent advancements in automatic calibration methodology have allowed parameters to be selected by optimizing single or even multiple objective functions (Madsen 2000).

Testing

Once the model is adequately calibrated, it is tested with a dataset not used during calibration, in a process called model testing. The same assessment methods are used as during model calibration. Once a certain level of fitness is reached, the model can be used for its intended purpose.

Once the model is adequately calibrated, it is tested with a dataset not used during calibration, in a process called model testing.

Model testing needs to be placed in context, because tested models cannot necessarily be transposed to a different climate or land use regime or a completely new geography. Even the most physically rigorous model can provide misleading results if used well outside its intended modelling space (Kuczera et al. 1993). Transposability, either in time or space, needs to be verified during the testing process (Klemeš 1986b, Refsgaard and Knudsen 1996). For example, if a model is to be used to assess the effects of drought, it should be calibrated against a wet period and tested during a dry period.

A tested model cannot necessarily be transposed to a different climate or land use regime or a completely new geography.

Almost all hydrological model calibration and testing is done at the catchment outlet using discharge as the assessment variable. This is completely acceptable if the purpose of the modelling exercise is to investigate water flows from a catchment. However, it leaves much to be desired if the goal of the modelling is to understand what is happening within the catchment.

Most hydrological model calibration and testing is done at the catchment outlet using discharge as the assessment variable. Ideally the catchment's internal workings would also be verified, but this is rarely done.

In an ideal world, the internal workings of a catchment are also verified. This could be accomplished using discharge measurements along the stream network (e.g., Sanford et al. 2007) or using spatial measurements of soil moisture or ET rates for example (e.g., Oudin et al. 2003), but the general lack of such data means that internal testing is rarely done. The use of remote sensing and distributed wireless measurement networks will go a long way in remedying this data gap.

6.2.4 Model evaluation

Error and uncertainty

No model output will perfectly match the observed measurements. The error contained in the output can be due to four different reasons (Butts et al. 2004):

- 1. random and systematic error in model inputs or boundary conditions;
- 2. random or systematic errors in outputs;
- 3. suboptimal parameterization; or
- 4. model bias.

It is imperative that some estimate of these errors be presented so that reasonable inferences can be made from the model output. At a minimum, measures of uncertainty can be computed by propagating values of the lower and upper bounds of parameter distributions through the model. There are also novel methodologies based on Monte-Carlo style likelihood estimation of parameter uncertainty (Beven and Binley 1992).

Most model users lack the expertise to alter a model's structure; the best chance at reducing error is to spend more time with parameterization and to ensure that the input data have the highest quality checks. For example, if the output is consistently biased, there could be some unrealistic parameter values.

The best chance at reducing error is to spending more time with parameterization and to ensure assuring that the input data have the highest quality checks.

Only model developers have the expert knowledge to tinker with the internal coding of the model to solve any random or systematic error in model output. They would also be able to add in processes/components that seem to be causing the high levels of error. Communication with model developers could be the most fruitful way forward if the modelling process stalls because of some systematic error, although this may not be possible for most managers. Also, communication between model users and model developers could be a good strategy to minimize error.

Equifinality

Even if error is low, models can still suffer from the bane of all modellers: equifinality. Equifinality occurs when multiple parameter sets give similar model output; this can undermine confidence in the entire model edifice. Does equifinality imply that parameter values are all fictional? What about the serious thought that goes into designing the models? It is a serious criticism that may not have a satisfactory solution. Nevertheless, modellers need to strive to get the right answers for the right reasons (Sayama and McDonnell 2009). Some modellers say that reducing equifinality is the best hope (Beven 2004).

Equifinality occurs when multiple parameter sets give similar model output; this can undermine confidence in the entire model edifice.

One way to reduce equifinality is to use other measures for model testing such as spatially distributed datasets, which can be used to narrow the possible set of parameter values (Franks et al. 1998). Another method is to use a handful of models and narrow parameter sets to look for convergence of model output (Georgakakos et al. 2004), but this is probably not a viable option for most forest managers due to its resource intensity.

A feaseble way to reduce equifinality is to use other measures for model testing such as spatially distributed datasets, which can be used to narrow the possible set of parameter values.

Further steps and options

If after going through the steps of evaluating a model as identified in Figure 17, an acceptable model accuracy is not reached, the modeller has a few options. First, more or better data can be acquired or errors can be corrected. Second, a different model can be chosen and the model selection steps repeated. For example, in larger drainage systems, models that include in-stream processing may be needed to better capture hydrological and biogeochemical processes.

If neither data nor models exist that can be used to answer the modelling question, the question itself needs to be changed or abandoned. While difficult to admit, modelling may be unfeasible in some instances. The uniqueness of place, non-linear dynamics, and threshold behaviour can throw curveballs to any wellintentioned modeller. Failure should be used as a learning exercise and may greatly inform hydrological modelling in general.

If neither data nor models exist that can be used to answer the modelling question, the question itself needs to be changed or abandoned. Modelling may be unfeasible in some instances and failure should be used as a learning exercise.

6.3 Models and forest management

Ecosystem management and the natural disturbance approach

Ecosystem management (EM) is a central tenet of a sustainable forest management approach in Canada. The primary goal of EM is to maintain the ecological integrity and health of the forest by managing the forest as an integrated system, within a conceptual framework designed to maintain ecosystem structure and function (Kimmins 2004, Gauthier et al. 2009).

The development of forest management practices that result in forest patterns and structure that more closely resemble those resulting from natural disturbance has been suggested by Hunter (1993) and others (Perera and Buse 2004, Bergeron and Harvey 1997, Attiwell 1994) as a means to achieve EM. This is best accomplished by designing management practices that result in forest conditions within the historic range of variability associated with natural disturbance regimes (Landres et al. 1999, Ontario Ministry of Natural Resources 2001; Andison 2003).

The development of forest management practices that result in forest patterns and structure that more closely resemble those resulting from natural disturbance has been suggested by experts as a means to achieve EM.

Managing hydrological systems; natural variability

Managing for a reference or natural condition has also been suggested for riverine systems (Poff et al. 1997), benthic macroinvertebrates (Reynoldson et al. 1997), in the biocriteria approach of the US Environmental Protection Agency (Davis and Simon 1995), and for biodiversity and ecosystem integrity of aquatic, wetland and riparian ecosystems (NRC 1992, Naiman et al. 2005). Similar to the natural-disturbance-based approach for managing forests described above, this paradigm stipulates that management should aim to maintain or restore the state (ecological and/or hydrological properties) of the hydrological system under management to a more natural condition that is chosen as a reference.

In the reference condition approach, a key concept is the importance of maintaining biological diversity and the variability inherent in natural systems.

A key concept of the reference condition approach is the importance of maintaining biological diversity and the variability inherent in natural systems. This is especially important when managing dynamic systems such as forests or hydrological networks. For example, using a one-time measurement of a hydrological property like river discharge is not appropriate; instead, a discharge time-series that defines the range of natural variation is needed (Poff et al. 1997).

However, when determining the range of natural variation the big question becomes: what is a realistic time period for calculating it? Some might say 10 to 20 years (Richter et al. 1997), but given some of the longer natural cycles such as the Pacific Decadal Oscillation, 100 years might be more appropriate.

Another factor to consider when using the range of natural variation as a management goal is the social implications of a given range of variability. Perhaps society is not willing to accept the range of conditions (e.g., low flows and floods) and we may wish to modify the range of natural variation used for the modelling exercise to more closely align with societal values (Klenk et al. 2009).

The need to establish the range of natural variation for hydrological and forest systems is one of the reasons long-term observation networks such as the LTER (Long Term Ecological Monitoring program) in the USA are crucial. While long-term measurements of watershed hydrological variables are extremely valuable for providing ranges of natural variability, most catchments are ungauged. Hydrological models are essential for defining the discharge reference conditions of ungauged systems over long periods of time. Once the reference conditions are defined, potential effects of various forest management scenarios may be simulated under current and projected climatic regimes.

What is a realistic time period for calculating the range of natural variation? What are the social implications of managing for a given range of variability? How are reference conditions defined? What are potential effects of different scenarios?

6.3.1 Using hydrological models to define the reference condition: needs and approaches

If the key to successful management is defining the natural or reference condition, then getting high quality model output that is right for the right physical reasons is a top research need. There has been a lot of effort by modellers to improve the characterization of hydrological processes. These include interception (Ellis et al. 2010), evapotranspiration (Running et al. 1989, Lu et al. 2003), snowmelt (Coughlan and Running 1997, Jost et al. 2009), and runoff processes (Beckers and Alila 2004).

There is still a big research gap in interfacing surface water and groundwater models, as both types of models treat each other as a boundary condition. A notable exception is InHM, which has fully integrated surface and subsurface processes (VanderKwaak 2000).

There is still a big research gap in interfacing surface water and groundwater models.

For subsurface pathways, the barrier to advancement is in large part due to inadequate characterization of the subsurface (Beven 2004). For surface pathways, the greatest shortcomings are related to the modelling of dynamic watershed objects such as lakes and wetlands, which expand and contract on a regular basis (Creed et al. 2002).

Incorporating our latest understanding of hydrology processes into hydro-ecological models is well ahead of the inclusion of biogeochemical processes. There is a growing recognition that the next generation of models will need to be integrated ecosystem models that consider the linkages between climate, hydrology and elemental cycling of nutrients. Appropriate modelling of vegetation dynamics is key to accurate forest hydrological simulation as forest composition changes and individual species adapt physiology to match the changing climatic conditions. The next generation of models will need to be integrated ecosystem models. They will need to consider the linkages between climate, hydrology and elemental cycling of nutrients.

One of the most advanced integrated modelling systems is a hierarchical regional ecosystem simulator: the Regional Hydro-Ecological Simulation SYStem (RHESSys). It consists of multiple submodels that simulate climate, water and biogeochemical cycling, with explicit consideration of scale in terms of process representation (Band et al. 1996, Tague and Band 2001a). RHESSys is one of the few models that can 'grow' forests and also allows for the simulation of forest management activities (Tague and Band 2001a) and fire spread (Krougly et al. 2009). Hwang et al. (2009) used RHESSys to demonstrate how forest ecosystems optimize carbon uptake by adapting physiological processes to make use of differential distribution of resources (water, light, and nutrients).

A few models can "grow" forests and allow for the simulation of forest management activities.

Hydrological connectivity: a dynamic property

As alluded to earlier, hydrological connectivity is the fundamental ecosystem property which is key to understanding hydrological and biogeochemical transfers between land and water and land and air. Hydrological connectivity is a dynamic property of ecosystems which controls the ease of movement of water, sediment and nutrients between different subcomponents of the system (Creed et al. 1996, Stieglitz et al. 2003, Istanbulluoglu et al. 2003). The successful simulation of hydrological connections is made difficult by the fact that many of these connections are controlled by non-linear and often threshold-dependent processes. Hydrological connectivity is a dynamic property of ecosystems which controls the ease of movement of water, sediment and nutrients between different subcomponents of the system.

Hydrological connectivity can be affected by many forest management activities (road construction, harvest and riparian management areas). It is the focus of many regulatory requirements and best management practices. A hydrological model that could assist with refining management prescriptions for these areas would be beneficial to forest managers and the ecosystems they manage.

Hydrological connectivity is key to understanding hydrological and biogeochemical transfers between land and water and land and air.

Challenges and future options

Modelling the hydrological reference condition for forested ecosystems is not easy, because watersheds are complex systems characterized by non-linear dynamics. Despite our best intentions, there is a tradeoff between more realistic representations of processes and parameter identifiability.

Perhaps the best way forward will involve the integration and improvement of measurement techniques (ground-based and remote), for spatial and non-spatial ecosystem characteristics that can be used to reduce parameter uncertainty. For example, soil moisture data from remote sensing have been used in hydrological models for improved flood prediction (Oudin 2003).

6.3.2 Modelling the impacts of forest management activities

6.3.2.1 Areal versus linear features (harvest blocks versus roads)

Forest harvesting can be a major agent of change in forests and has important effects on downstream water quantity and water quality. Harvesting changes forests by removing vegetation, altering the age class distribution and composition, and disturbing the soil. These changes can alter ecological processes and functions and lead to loss of biological integrity. Extensive research and monitoring has been conducted on the hydrological effect of forest harvest, which in most places is accomplished using clear cut silvicultural systems (Hornbeck et al. 1993, Stednick 1996, Bonell 1998).

However, research has found that the effect of roads on modifying the hydrology of forests can be just as important as harvest-related effects (Tague and Band 2001b). Roads change the surface hydrological connectivity by channelling water and the associated sediments and nutrients along hard packed surfaces or associated ditches. Road networks may have a stronger influence on peak flows than harvest areas (Eisenbies et al. 2007). In fact, the environmental effects of roads in some forest areas have prompted the Environmental Protection Agency in the USA to consider reclassifying forest roads as point sources of pollution (Eisenbies et al. 2007).

Hydrological models can be used to simulate the possible impacts of harvesting activities, although their full potential has not been realized since many of the simulations have been done on a lumped basis. Only a few hydrological models have addressed road impacts (Table 11).

Forest harvest and roads both have important effects on hydrology and water quality. Models can be used to simulate such impacts, but their full potential has yet to be realized.

6.3.2.2 Scaling: from headwater catchments to larger basins

Much of our understanding of hydrological processes comes from studies conducted in small headwater catchments (first- to second-order basins). This is also true of paired-basin field studies investigating the effects of forest management. Such paired-basin studies can only be conducted in low order catchments where precipitation inputs, soil and geologic conditions, topography, and other variables are the same in the treatment and control catchments (Bonell and Bruijnzeel 2004). In higher catchment orders, the sources of variation in physical catchment characteristics increase substantially and control catchments are hard to find. At higher scales, virtual catchments created using modelling can fill the role of control watersheds (Eisenbies et al. 2007).

Virtual catchments created using modelling can fill the role of control watersheds.

It is important to consider what happens to water flow when catchment scale is increased. With respect to the integrated response at the outlet, there will be a lot of mixing of the different source waters. This is the reason some of the effects of forest management activities in low order catchments end up being averaged away (although there is probably a threshold in disturbance severity beyond which impacts are transferred downstream). Also, at higher orders, in-stream processes increase in importance and need to be represented in hydrological models.

Despite these concerns, Merz et al. (2009) found that many model parameters did not change with spatial scale. They found that some of the differences in model parameters had less to do with scale than with differences in geology and other physical characteristics. This is a challenge because as scale increases, the spatial heterogeneity also increases and the resultant dynamics may have as much to do with scale as with biophysical differences. Merz et al. (2009) also found that water balances can be modelled more reliably at larger scales than at finer scales. At larger scales, response at the outlet reflects mixing of different source waters. This tends to "average away" some of the effects of forest management.

An innovative use of models at higher scales is change detection modelling where different model runs are made using the split sample technique and model parameters and output are analyzed to detect any changes (Seibert and McDonnell 2010). The aim is to define a representative modelling unit for which statistical description will suffice. The search for this representative unit is ongoing with some fruitful advances (e.g., Sanford et al. 2007).

6.3.2.3 Single versus cumulative impacts

In today's rapidly changing world, forests are subjected to multiple and repeated impacts. The effects of these impacts are cumulative and need to be assessed using a combination of field monitoring and numerical modelling. It is important to understand the effects of past land management on current hydrology and to separate them from the effects of the current environmental (and management) conditions they interact with.

It is important to understand the effects of past land management on current hydrology and to separate them from the effects of the current environmental (and management) conditions they interact with.

The key to understanding cumulative effects is the analysis of long-term datasets, since cumulative effects only manifest themselves after many years. Models can be used to extend past measurement records, and to forecast possible futures under changing environmental and climatic regimes as well as different management scenarios. Linking hydrological models to spatial forest planning tools can facilitate the selection of management options that will minimize effects on water resources. The key to understanding cumulative effects is the analysis of long-term datasets. Models can be used to extend past records and to model future scenarios.

There are now a multitude of studies modelling the interactive effects of land-use and climate change (Istanbulluoglu et al. 2004, Ma et al. 2010). Given the complexity of ecosystem behaviour, the need for large datasets and expert knowledge requirements, cumulative effects are best analyzed by a team of scientists (e.g., FORWARD project, Putz 2003).

6.3.2.4 Current versus past impacts

Hydrologists have recognized that a landscape's current hydrological state is in part a function of past land uses; this is called the legacy effect (Foster et al. 2003, Jones et al. 2009). For example, the practice of clear-cutting and fire-suppression has changed the forest structure, species composition, and related vulnerability to insect infestation. Moreover, many agricultural landscapes on the eastern seaboard have reverted to forests, but their current state is now also influenced by past land use which, for example, had changed the soils and the seedbank.

A landscape's current hydrological state is in part a function of past land uses.

6.3.2.5 Modelling water yield and water quality in ungauged catchments

It is challenging to construct physically realistic models that account for land-use change, and that can be applied to ungauged catchments both in temperate and tropical regions. There are a number of promising, mostly lumped models that have been used in ungauged basins to simulate water yield.

Water yield

One model (SWAT, Arnold et al. 1998) has been effectively used to model the effect of land-use change (Govender and Everson 2005, He et al. 2008, Alansi et al. 2009). One of the key strengths of SWAT lies in its ability to model the relative impacts of changes in management practices, climate and vegetation on water quantity and quality. Other lumped models include HYLUC (Calder 2003), IHACRES (Croke and Jakeman 2003), and WRENSS (US EPA 1980). Some of these were specifically designed to estimate the change in annual water yield following a variety of silvicultural prescriptions (Swanson 1998).

SWAT is a model that effectively represents the effect of land-use change.

Water quality

Modelling downstream water quality has also been mostly done with lumped hydrological models connected to a GIS-based nutrient export model. For example, El-Kaddah and Carey (2004) used HSPF and a non-point source model (NPSM) integrated within a GIS-based software package. While they were successful at modelling water flows there was poor agreement with total nitrogen concentrations, which they attributed to the lack of good input data and the lack of nitrogen transformation processes built into NPSM. Francos et al. (2001) linked SWAT with a GISbased water quality model to characterize a partly agricultural and partly forested landscape in Finland.

Modelling downstream water quality has also been mostly done with lumped hydrological models connected to a GIS-based nutrient export model.

Changing land uses, changing models

Mixed land use forests will be more widespread in the future, as agriculture and urbanization move into forested areas. Hence, models need to be able to handle multiple land uses to be effective tools. The bottom line will be getting the hydrology right, since the timing and magnitude of the flow determines the export. Lindgren et al. (2007) reported large differences between two models simulating total nitrogen export, mostly due to spatial variation of specific runoff. Some people have abandoned process-based modelling altogether due to the difficulty of capturing the necessary biogeochemical processes. Purely statistical techniques such as artificial neural networks and similar approaches (Nour et al. 2006) are gaining in popularity.

Models need to be able to handle multiple land uses to be effective tools.

6.4 Modelling the future of water resources under a changing climate

Global circulation models (GCMs) are routinely used to predict the future climate with increasing accuracy regarding global average temperature and precipitation. However, the local manifestation of GCM results, especially in terms of water, is highly variable and debated. A large scientific effort is underway to try to bridge the gap between predictions at the scale of GCM cells and the scales necessary for planning and management at regional and local levels (Fowler et al. 2007). Various statistical or dynamic downscaling methods have been proposed. An example of a dynamic model is the Vegetation/Ecosystem Modelling and Analysis Project (VEMAP) (Kittel et al. 1995). It has produced a database of precipitation and daily temperature maxima and minima at a 0.5° grid resolution for the contiguous USA between 1895 and 2100.
The magnitude, timing, and variability of hydrological fluxes may all change, with various effects on the ecology of forests and their waters. Vegetation changes can in turn have large effects on evapotranspiration.

The challenge of developing process-based models for predicting effects of climate change is that ecosystem processes (physical, chemical and biological) will also be changing. As a consequence we need models that can respond to these changing conditions. Pike et al. (2008) listed eight key hydrological consequences of climate change, including:

- 1. increased atmospheric evaporative demand;
- 2. altered vegetation composition affecting evaporation and precipitation interception;
- 3. decreased snow accumulation and earlier melt;
- 4. accelerated melting of permafrost, lake ice, and river ice;
- 5. glacier mass balance adjustments;
- 6. increased stream and lake temperatures;
- 7. increased frequency or magnitude of disturbances; and
- 8. altered streamflow.

The magnitude, timing, and variability of hydrological fluxes may all change. This will have various potential effects on the ecology of forests and their waters.

The models of the future need to be able to address these changes. Local and regional models need to consider changes in vegetation, which will have very large impacts on evapotranspiration. Currently only RHESSys is capable of representing changes in the physiology of plants (Tague and Band 2004).

For cold-water hydrology, modelling freeze/thaw dynamics of glaciars and permafrost will be critical.

Runoff modelling will need to represent properly preferential flow through macropores, along bedrock/soil interfaces and especially along roads, because these objects significantly influence quick flows leading to extreme downstream flooding (Beckers and Alila 2004). Part of this modelling process will require determining the thresholds at which preferential flowpaths become activated (Tromp-van Meerveld and McDonell 2006).

In terms of cold-water hydrology, modelling the freeze/thaw dynamics of glaciers and permafrost will be critical. These dynamics have huge consequences for downstream freshwater ecosystems, including the human components (Loukas et al. 2002, Ma et al. 2010). They also have a huge potential to release vast quantities of methane hydrates (Mascarelli 2009).

6.5 Barriers to the operational use of models

Models are useful because of their ability to simulate future watershed conditions under varying management practices and changing climate (Beckers et al. 2009). However, hydrological models are not widely used by forest managers. This is due to several barriers which need to be overcome, from a scientific rather than management perspective.

6.5.1 Bridging the gap between hillslope complexity and watershed simplicity

Sivapalan (2003) argued that while we have greatly advanced our understanding of process complexity at the hillslope scale, we have not been able to translate this understanding to explain why things are much simpler to model and predict at the watershed scale. In building parsimonious watershed models, the right balance must be struck with keeping the necessary detail learned at the hillslope scale.

Given the fact that location of forest management activities matters, and spatial modelling is now widely used in forest management planning, the use of spatially distributed hydrological models will be necessary. This implies a need for fairly detailed descriptions of processes at the hillslope scale. Given the fact that location of forest management activities matters, and spatial modelling is now widely used in forest management planning, the use of spatially distributed hydrological models will be necessary.

6.5.2 Incorporating field and remote sensing measurements in hydrological models

Models need field measurement for parameterization, calibration and validation. Most commonly the only type of measurement used by modellers is discharge obtained at the outlet of a catchment. However, field experimentalists measure a lot more than discharge. While datasets on groundwater head values or soil moisture are currently difficult to acquire, new ground-based (via wireless sensors) and remote sensing (RADAR, LiDAR, and Hyperspectral) datasets will provide modellers with much more than just discharge data. Reconciling field measurement with model output will most likely require the incorporation of tricky non-linear and threshold-type dynamics.

Models need field measurement for parameterization, calibration and validation. Field and remote sensing datasets can also be incorporated into model functioning.

The incorporation of field and remote sensing datasets might take different forms. Such datasets can be used solely to calibrate models and test model output, but they can also be readily incorporated into the functioning of the model. For example, hourly or daily remote sensing data on soil moisture might be fed into the model to update its water accounting. Moreover, remote sensing can provide frequent updates to vegetation parameters which are unrealistically treated as static in most models.

6.5.3 Integrating models at different scales

Integration of models from different scales needs to be strengthened. For example, future global climate is best modelled using global circulation models. However, local and regional hydrological predictions need to be made with models at finer scales. The global climate information has to be downscaled for the regional model, which can be further downscaled to local (watershed) models.

We have pointed out the differences in processes that can exist between hillslope and catchment scales; similar differences exist between headwater catchment models and the more regional models preferred by forest managers. This is why studying scaling in the treatment of cumulative and interactive processes is so important.

Differences exist between headwater catchment models and the more regional models preferred by forest managers. Integration of models from different scales needs to be strengthened.

If forest managers are to design and implement an integrated systems approach for both aquatic and forested ecosystems, a method to model processes at multiple scales is needed. For instance, the design of riparian buffers is generally done at a local scale, based on field inspections and aerial photographs, with little regard for activities up or down stream, or at a catchment or watershed scale.

In order to manage hydrological systems effectively at multiple scales, within an ecosystem management framework, integrated models are needed (Donnelly 2003). To truly integrate riparian management strategies at the stand scale with strategies at the landscape level (for instance regarding cumulative harvest or disturbance levels), we need to be able to predict the effects of management scenarios at multiple scales over the length of a planning period (generally 100 years). Strategic forest management models (e.g., PATCH-WORKS, Spatial Planning Systems 2008) are now able to forecast future landscape conditions spatially, based on a wide variety of variables and indicators. The ability thus exists to forecast the cumulative effects on water resources by using hydrological models that can incorporate the model output (forest landscapes) projected through time from a forest management model.

Strategic forest management models are able to forecast the cumulative effects on water resources by using hydrological models that can incorporate the model output projected through time from a forest management model.

6.5.4 Classifying models based on purpose and hydrological processes

The need for catchment classification is great. It would greatly aid model and parameter selection for ungauged catchments. A catchment classification would help to determine which models should be used where and when, and would readily lend itself to the development of guidelines for planning and operational practices specified for different types of catchments.

A catchment classification would help to select the best model and parameters for ungauged catchments.

6.6 Conclusions

Models are human constructs based on limited knowledge of how the 'real' world works, yet they give valuable insight for science and decision making. Given the global scale changes in climate and associated ecosystem processes such as fires, insect infestations and species migration, water resources will be greatly diminished for both ecosystem and human use. Hydrological models give us a tool that can help us evaluate how current trends and actions will influence the future.

Modelling will help us manage sustainably for multiple ecosystem services.

Scenario modelling is very useful in determining the best way forward. There are multiple barriers related to model parameterization, data collection, and process representation, but pooling resources between communities, agencies, academics and corporations will unlock the power of the collective to solve problems together.

We need to manage for multiple ecosystem services in order to achieve sustainability. Models will help in our efforts to implement adaptive management. Complexity in systems is inherent - we need to deal with it and see the forest through the trees.

Pooling resources between communities, agencies, academics and corporations will unlock the power of the collective to solve problems together.

SCIENTIFIC THEORY, DATA AND TECHNIQUES FOR CONSERVATION OF WATER RESOURCES WITHIN A CHANGING FORESTED LANDSCAPE IRENA CREED ET AL. 2011 | STATE OF KNOWLEDGE REPORT | SUSTAINABLE FOREST MANAGEMENT NETWORK

Recommendations and conclusions

Field, remote sensing, and simulation-based methods for understanding forest hydrology have produced a substantial amount of information on water resources of Canada's forests. Some of this information has already been mined by scientists, and the knowledge generated has been implemented by forest managers. There is still considerable knowledge to be gained in organizing, analyzing and synthesizing the existing databases, and translating the knowledge into effective decision making. However, to improve the production of knowledge it is important to extend our observations to areas of Canada's forests that have not been adequately sampled to ensure that we have the understanding required to conserve water resources in all of Canada's forested landscapes.

In this SOK report, we have pointed out important gaps in scientific knowledge as well as the limits of the tools detailed. None of these tools offer a panacea in terms of understanding watershed behaviour. They all have their strengths and weaknesses, and in the final analysis they all rely on each other. In conclusion, we advocate a strategy based on expanding data collection using appropriate measurement tools, followed by fusion and assimilation of these new data with existing databases. This will allow to produce data products that can be readily accessed and analyzed by managers. The future is all about integrating and sharing data and knowledge.

The following subsections highlight the main recommendations for scientists, managers and policy makers to improve access to hydrological data, tools and techniques for effective management of water resources in our changing forested landscapes.

7.1 Create a national data archive for forest hydrology

It is important to continue research projects that seek to compile hydrological information nation-wide, such as the SFMN funded HELP project. It is also important to build a national forest hydrology web portal to bring together all the relevant datasets (WSC and research gauging stations, MSC meteorological data, field-based monitoring of soil moisture and other parameters, remote sensing) into a single, publicly accessible venue.

Currently, the HELP database and webportal are housed by CFS (http://canforhydro.org); renewed support could potentially grow this webportal/database into the needed Canadian warehouse for hydrological information. Such a centralized database would encourage value-added cross site comparisons and meta-experiments, greatly expand knowledge generation, and increase the incorporation of these resources into forest management planning and operations. This centralized web portal should also be linked to global datasets for ease of upscaling or downscaling hydrological information. Access is key to quick dissemination of information, and therefore we advocate the American model of freely sharing tax-payer funded datasets.

7.2 Continue existing and establish new monitoring networks

In this digital age of satellite sensors and distributed sensor networks, field based measurements of hydrological parameters are still crucial. Adequate sampling of hydrological parameters across the variety of hydrogeological systems present in Canada is necessary for understanding Canadian hydrological regimes, and to the development of accurate hydrological models. In general, Canada has lost many meteorological and water survey stations since the golden age of hydrometric data collection in the 1970s. This trend of closures, however, needs to be stopped and reversed. Ground based sensors cannot be eliminated, because they are key to measuring variables like discharge, deeper soil storage, and vertical profiles of snowpack temperatures. The existing hydrometric stations need to be maintained because many of the critical questions, especially those related to climate change, can only be addressed by analyzing long-term datasets. Canada needs to take a lesson from the USA where there is a whole network of long-term observation sites such as Long-Term Ecological Research [LTER], National Ecological Observatory Network [NEON], and Consortium of Universities for the Advancement of Hydrologic Science [CUAHSI].

In addition to continuing data collection at existing stations, the monitoring network needs to be expanded to stations targeting meso-scale watersheds, which size is more compatible with management activities. Government agencies should take the lead in developing a nationally coordinated ecohydrology monitoring network, and build partnerships with universities and local stakeholders, including the public, to increase monitoring of the environment. A national ecohydrology network will be crucial if Canada is to sustainably manage it's most strategic and valuable resource (water), in the face of increasing demands and increasing threats, such as climate change and assorted human activities. Other countries have supported and expanded their monitoring networks to enhance the protection and management of their national natural resources. Canada risks becoming irrelevant in international arenas if we cannot bring adequate data and knowledge of the natural resources under our stewardship to global climate change and adaptation discussions.

7.3 Encourage and adopt emerging technologies for quantitative analysis

We need improved sensors for surface and subsurface *in-situ* sampling and for airborne and satellite remote sensing of various hydrological phenomena. Ideally most of the new ground based sensors should be cost effective and easier to use in order to cover multiple key points in different sized watersheds. In terms of remote sensing sensors, the hydrological community needs, most of all, microwave sensors with multiple polarizations and frequencies. These will allow enhanced vegetation canopy penetration and better detection of soil moisture and snow water equivalent. In regions where groundwater contributions are an important component of the water balance, the development of new geophysical imaging systems along the lines of the GRACE sensor (which measures total water content) is crucial. We need sensors that can map the soil-bedrock interface, the critical water holding (aquifers), and conducting layers (sand deposits), to increase our ability to characterize the subsurface. In order to effectively image the surface of Canada's forests under the thickest of canopies, a satellite based and high spatial resolution (\approx 5m) LiDAR sensor is desperately needed as well.

The integration of remote sensing imagery in hydrological models will continue to be a critical research area for years to come. Further integration of digital terrain analysis, remote sensing, and distributed simulation modelling data sets with GIS-based spatial decision support systems will be especially important for managers and other decision makers. The transfer of statistical techniques from other fields (e.g., spectral analysis) has introduced significant opportunities for research in applying these methods to hydrological applications. Given the broad geographic variation of Canada's hydrological landscapes, the classification of suitable measurement and analytical tools for different locations or geographic areas is required. The digital toolkit, mentioned in section 7.2, will not achieve its fullest capabilities unless it is supported by comprehensive ground-based networks.

7.4 Promote integration of datasets at relevant scales

Hydrological scientists are currently focusing their efforts on research at the scale of headwater catchments (≈ 1 km²), which is disconnected from the scale of management which occurs over drainage basins (≈ 100 s km²). If forest managers are to design and implement an integrated systems approach for managing both aquatic and forested ecosystems, a method to model processes at multiple scales over the length of a planning period is needed. We need to spatially integrate explicit forest modelling processes and hydrological tools, to enhance our ability to upscale results from headwater catchments to regional drainage basins, consider in-stream, lake and wetland processes. This integration will allow the forecasting of cumulative effects of forest management activities on water resources through time. In terms of predicting the effects of forest management under a changing climate, the global climate circulation models need to be downscaled from the currently available regional levels to local (watershed) levels.

The integrated modelling systems approach to forest hydrology consists in integrating the different types of data reviewed in this report. This system would embrace the strengths of hydrometric observatories, distributed wireless sensors, ground based radars, remote sensing systems, geophysical sensors, and hydrological models (static and dynamic). The measurements and modelled output can be compiled in a GIS setting and broadcasted on the internet. An example of such integrated modelling system is the use of static maps of digital terrain analysis, which can form the basis for dynamic modelling of processes and patterns that are calibrated by local and ground-based monitoring networks. Together with remote sensing data layers, such as soil moisture and snow cover, model predictions can be constrained to give results with acceptable uncertainty bounds. These results could be useful to managers in making decisions such as road and cutblock location and design, watershed-level harvest targets, riparian management strategies, and forest renewal prescriptions. These catchment scale analysis systems should be freeware with transparent code, like the Terrain Analysis System and its successor Whitebox (Lindsay 2005).

7.5 Develop a national watershed classification system

An ecohydrology-based watershed classification system is a critical tool for researchers, forest managers and policy makers when developing management guidelines and operational practices. The availability of such a tool would enable planners to recognize the spatial variability within and between different catchments, allowing the refinement of forest management prescriptions based on site conditions and associated hydrological processes. It would also allow to minimize the potential for adverse effects from harvest or renewal activities. For researchers, a classification system would provide a framework for codifying results and developing new knowledge of hydrological processes across scales. A watershed classification would greatly aid model and parameter selection for ungauged catchments, and help determine which models should be used where and when. The classification would also show where the gaps in our scientific understanding and management consideration are.

7.6 Promote hydrological training and knowledge transfer

Even if the best of data, tools and models are available, they will languish if there are not qualified people to use them. In an era of increasing water insecurity, it is essential to promote and enhance interdisciplinary training in water resource sciences and applications. Forest hydrologists have been slow to adopt and apply the novel hydrological techniques to forest management planning presented in this report. We need enhanced knowledge transfer from product developers in digital terrain analysis, remote sensing, and distributed simulation modeling, to forest hydrologists and managers. It is important that forest managers and scientists work together to better understand the complexity of the management planning, because it helps to reduce uncertainty and risks in terms of forest sustainability, economic and social values. Uncertainty, error and caveats associated with the latest products need to be well documented, otherwise their use will not be universal. The users of spatial datasets will need to learn how to manage and process raw data, turn it into information, and then transform it into know-ledge. An encouraging way forward is by using Web 2.0 technologies, where information flows both ways and "end-users" become "engaged-users".

In summary, meeting the coming challenges to forest water resources (increased demands, climate change and sustainability) requires improved data, new tools and the qualified people to integrate them into forest management. A revitalized forest hydrology community (industry, governments, academia and citizens) equipped with better data, tools and knowledge will ensure the future health of forest ecosystems, communities and the forest industry, and will demonstrate that we are responsible stewards of a significant proportion of the world's forest and water resources.

8.0

References cited

- Adams, K. N. and Fowler A. M. 2006. Improving empirical relationships for predicting the effect of vegetation change on annual water yield. Journal of Hydrology 321: 90-115.
- Agnew, L. J., Lyon, S., Gerard-Marchant, P., Collins, V. B., Lembo, A. J., Steenhuis, T. S. and Walter, M. T. 2006. Identifying hydrologically sensitive areas: Bridging the gap between science and application. Journal of Environmental Management 78: 63-76.
- Alansi, A. W., Amin, M. S. M., Halim, G. A., Shafri, H. Z. M. and Aimrun, W. 2009. Validation of SWAT model for stream flow simulation and forecasting in Upper Bernam humid tropical river basin, Malaysia. Hydrology and Earth System Sciences Discussions 6: 7581-7609.
- Alsdorf, D. E., Birkett, C., Dunne, T., Melack, J. and Hess, L. 2001. Water level changes in a large Amazon lake measured with spaceborne radar interferometry and altimetry. Geophysical Research Letters 28: 2671-2674.
- Alsdorf, D. E. and Lettenmaier, D. P. 2003. Tracking fresh water from space. Science 301: 1491-1494.
- Alsdorf, D. E., Melack, J. M., Dunne, T., Mertes, L. A. K., Hess, L. L. and Smith, L. C. 2000. Interferometric radar measurements of water level changes on the Amazon flood plain. Nature 404: 174-177.
- Andison, D.W. 2003. Tactical forest planning and landscape design. In Towards sustainable management of the boreal forest, P.J. Burton, C. Messier, D.W. Smith and W.L. Adamowicz (eds.). Ottawa, NRC Research Press. 443-480.
- Andreadis, K. M., Liang, D., Tsang, L., Lettenmaier, D. P. and Josberger, E. G. 2008. Characterization of errors in a coupled snow hydrology-microwave emission model. Journal of Hydrometeorology 9: 149-164.
- Arnold, J. G., Srinivasan, R., Muttiah, R. S. and Williams, J. R. 1998. Large area hydrological modeling and assessment Part 1: Model development. Journal of the American Water Resources Association 34: 73-89.
- Attiwell, P.M. 1994. The disturbance of forest ecosystems: the ecological basis for conservative management. Forest Ecology and Management 63: 247-300.
- Avissar, R., Silva Dias, P. L., Silva Dias, M. A., Nobre, C. A. 2002. The Large Scale Biosphere Atmosphere Experiment in Amazonia (LBA). Insights and future research needs. Journal of Geophysical Research 107.
- **Baker, M. E., Wiley, M. J., Carlson, M. L. and Seelbach, P. W. 2003**. A GIS model of subsurface water potential for aquatic resource inventory, assessment, and environmental management. Environmental Management 32: 706-719.

- Band, L. E., Mackay, D. S., Creed, I. F., Semkin, R. and Jeffries, D. 1996. Ecosystem processes at the watershed scale: Sensitivity to potential climate change. Limnology and Oceanography 41: 928-938.
- **Band, L. E. and Moore, I. D. 1995**. Scale-landscape attributes and geographical information-systems. Hydrological Processes 9: 401-422.
- Bastiaanssen, W. G. M., Menenti, M., Feddes, R. A. and Holtslag, A. A. M. 1998. A remote sensing surface energy balance algorith for land (SEBAL). 1. Formulation. Journal of Hydrology 212: 198-212.
- Batelaan, O. and De Smedt, F. 2007. GIS-based recharge estimation by coupling surface-subsurface water balances. Journal of Hydrology 337: 337-355.
- Batelaan, O., De Smedt, F., Valle, M. N. O. and Huybrechts, W. 1993. Development and application of a groundwater model integrated in the GIS GRASS. In Application of geographic information systems in hydrology and water resources management, K. Kovar and H. P. Nachtnebel (eds.). 581-589.
- Becker, M. W. 2006. Potential for satellite remote sensing of ground water. Ground Water 44: 306-318.
- **Beckers, J. B. and Alila, Y. 2004**. A model of rapid preferential hillslope runoff contributions to peak flow generation in a temperate rain forest watershed. Water Resources Research 40: W03051.
- **Beckers, J. B., Smerdon, B. and Wilson, M. 2009.** Review of hydrological models for forest management and climate change applications in British Columbia and Alberta. FORREX Series 25, Forum for Research and Extension in Natural Resources Society, Kamloops BC, 179 pp.
- **Bent, G. C. 2001**. Effects of forest-management activities on runoff components and ground-water recharge to Quabbin Reservoir, central Massachusetts. Forest Ecology Management 143: 115-129.
- **Bergeron, Y. and Harvey, B. 1997.** Using silviculture on natural ecosystem dynamics: an approach to the southern boreal mixedwoods of Quebec. Forest Ecology and Management 92: 235-242.
- Best, A., Zhang, L., McMahon, T. Western, A. and Vertessy, R. 2007. A critical review of paired catchment studies with reference to seasonal flows and climatic variability. Murray Darling Basin Commission Publication 11/03, Murray-Darling Basin Commission, Canberra, ACT, Australia.
- **Beven, K. J. 1986**. Runoff production and flood frequency in catchments of order n: an alternative approach. In Scale problems in hydrology V.K. Gupta, I. Rodriguez-Iturbe and E.F. Wood (eds), D Reidel, Dordrecht, 107-131.
- **Beven, K. J. 1993**. Prophecy, reality and uncertainty in distributed hydrological modeling. Advances in Water Resources 16: 41-51.
- Beven, K. J. 1997. TOPMODEL: A critique. Hydrological Processes 11: 1069-1085.
- Beven, K. J. 2000. On the future of distributed modelling in hydrology. Hydrological Processes 14: 3183-3184.
- **Beven, K. J. 2004**. Towards environmental models of everywhere: advances in modelling and data assimilation. Proceedings of the BHS International Hydrology Symposium. Hydrology: science and practice for the 21st century, 244-250.
- Beven, K. J. 2008. On doing better hydrological science. Hydrological Processes 22: 3549-3553.
- **Beven, K. J. and Binley, A. 1992**. The future of distributed models model calibration and uncertainty prediction. Hydrological Processes 6: 279-298.
- **Beven, K. J. and Kirkby, M. J. 1979**. A physically based, variable contributing area model of basin hydrology. Hydrological Sciences - Bulletin - des Sciences Hydrologiques 24: 43-69.

- Bishop, K., Buffam, I., Erlandsson, M., Folster, J., Laudon, H., Seibert, J. and Temnerud, J. 2008. Aqua incognita: the unknown headwaters. Hydrological Processes 22: 1239-1242.
- **Blöschl, G. and Sivapalan, M. 1995**. Scale issues in hydrological modeling a review. Hydrological Processes 9: 251-290.
- **Blumenfeld, S., Lu, C., Christophersen, T. and Coates D. 2009**. Water, Wetlands and Forests. A review of ecological, economic and policy linkages. Secretariat of the Convention on Biological Diversity and Secretariat of the Ramsar Convention on Wetlands, Montreal and Gland. CBD Technical Series.
- Bobba, A. G., Bukata, R. P. and Jerome, J. H. 1992. Digitally processed satellite data as a tool in detecting potential groundwater-flow systems. Journal of Hydrology 131: 25-62.
- **Bonell, M. 1998**. Possible impacts of climate variability and change on tropical forest hydrology. Climatic Change 39: 215-272.
- **Bonell, M. 2005**. Runoff generation in tropical forests. In Forests, Water and People in the Humid Tropics, M. Bonell and L. A. Bruijnzeel (eds.) UNESCO Cambridge University Press, Cambridge, UK.
- **Bonell, M. and Gilmour, D. A. 1978**. The development of overland flow in a tropical rainforest catchment. Journal of Hydrology 39: 365-382.
- **Bonell, M. and Bruijnzeel, L.A. 2004**. Forests, water and people in the humid tropics: past, present and future hydrological research for integrated land and water management. Cambridge University Press. 906 pp.
- Borga, M., Fontana, G.D. and Cazorzi, F. 2002. Analysis of topographic and climatic control on rainfall-triggered shallow landsliding using a quasi-dynamic wetness index. Journal of Hydrology 268: 56-71.
- **Borwick, J., Buttle, J. and Ridgway, M. S. 2005.** A topographic index approach for identifying groundwater habitat of young-of-year brook trout (*Salvelinus fontinalis*) in the land–lake ecotone. Canadian Journal of Fisheries and Aquatic Sciences 63: 239–253.
- **Bosch, J. M. and Hewlett, J. D. 1982.** A review of catchment experiments to determine the effect of vegetation changes on water yield and evapo-transpiration. Journal of Hydrology 55: 3-23.
- **Bourgine, B. and Baghdadi, N. 2005**. Assessment of C-band SRTM DEM in a dense equatorial forest zone. Comptes Rendus Geoscience 337: 1225-1234.
- **Bourque, C. P.-A. and Pomeroy, J. H. 2001**. Effects of forest harvesting on summer stream temperatures in New Brunswick, Canada: an inter-catchment, multiple-year comparison. Hydrology and Earth System Sciences 5: 599-613.
- **Bracken, L. J. and Croke, J. 2007**. The concept of hydrological connectivity and its contribution to understanding runoff-dominated geomorphic systems. Hydrological Processes 21: 1749-1763.
- Brais, S., Paré, D., Camiré, C., Rochon, P. and Vasseur C. 2002. Nitrogen net mineralization and dynamics following whole-tree harvesting and winter windrowing on clayed sites in northwestern Quebec. Forest Ecology and Management 157: 119-130.
- **Bruijnzeel, L. A. 1988**. (De)forestation and dry season flow in the tropics: a closer look. Journal of Tropical Forest Science 2: 229-243.
- **Bruijnzeel, L. A. 2004.** Hydrological functions of tropical forests: not seeing the soil for the trees? Agriculture, Ecosystems and Environment 104: 185-228.

- **Bulcock, H. H. and Jewitt, G. P. W. 2010.** Spatial mapping of leaf area index using hyperspectral remote sensing for hydrological applications with a particular focus on canopy interception. Hyrology and Earth System Sciences 14: 383-392.
- Burns, D. A., Plummer, L. N., McDonnell, J. J., Busenberg, E., Casile, G. C., Kendall, C., Hooper, R. P., Freer, J. E., Peters, N. E., Beven, K. and Schlosser, P. 2003. The geochemical evolution of riparian ground water in a forested piedmont catchment. Ground Water 41: 913-925.
- **Buttle, J. M. 2002**. Rethinking the donut: the case for hydrologically relevant buffer zones. Hydrological Processes 16: 3093-3096.
- Buttle, J. M., Creed, I. F. and Moore, R. D. 2005. Advances in Canadian forest hydrology, 1999-2003. Hydrological Processes 19: 169-200.
- Buttle, J. M., Creed, I. F. and Moore, R. D. 2009. Advances in Canadian forest hydrology, 2003-2007. Canadian Water Resources Journal 34: 113-125.
- Buttle, J. M., Creed, I. F. and Pomeroy, J. W. 2000. Advances in Canadian forest hydrology, 1995-1998. Hydrological Processes 14: 1551-1578.
- Buttle, J. M. and Murray, C. D. 2007. Review of research on the forest-water interface in Canada: Report prepared for the Canadian Forest Service. 38 pp.
- **Buttle, J. M. and Eimers, M. C. 2009**. Scaling and physiographic controls on streamflow behaviour on the Precambrian Shield, south-central Ontario. Journal of Hydrology 374: 360-372.
- Buttle, J. M. and Metcalfe, R. A. 2000. Boreal forest disturbance and streamflow response, northeastern Ontario. Canadian Journal of Fisheries and Aquatic Sciences 57: 5-18.
- Butts, M. B., Payne, J. T., Kristensen, M. and Madsen, H. 2004. An evaluation of the impact of model structure on hydrological modelling uncertainty for streamflow simulation. Journal of Hydrology 298: 242-266.
- **Bwangoy, J. R. B., Hansen, M. C., Roy, D. P., De Grandi, G. and Justice, C. O. 2010**. Wetland mapping in the Congo Basin using optical and radar remotely sensed data and derived topographical indices. Remote Sensing of Environment 114: 73-86.
- **Calder, I. R. 2003**. Assessing the water use of short vegetation and forests: Development of the Hydrological Land Use Change (HYLUC) model. Water Resources Research 39: 1318.
- **Canadian Council of Forest Ministers. 1996.** Criteria and indicators of Sustainable Forest Management in Canada. Technical Rpt. Nat. Res. Canada, Cdn Forest Serv. Ottawa, ON.
- **Carginan, R. and Steedman, R. J. 2000**. Impacts of major watershed perturbations on aquatic ecosystems. Canadian Journal of Fisheries and Aquatic Sciences 57 (Suppl 2): 1-4
- **Chanasyk, D. S., Whitson, I.R., Mapfumo, E., Burke, J. M. and Prepas, E. E. 2003**. The impacts of forest harvest and wildfire on soils and hydrology in temperate forests: a baseline to develop hypotheses for the Boreal Plain. Journal of Environmental Engineering and Science 2: 51-62.
- Chang, M. 2003. Forest hydrology: an introduction to water and forests. CRC Press, Boca Raton, FL.
- **Chen, B. Z. and Coops, N. C. 2009**. Understanding of Coupled Terrestrial Carbon, Nitrogen and Water Dynamics-An Overview. Sensors 9: 8624-8657.
- **Clark, R. B., Creed, I. F. and Sass, G. Z. 2009.** Mapping hydrologically sensitive areas on the Boreal Plain: a multitemporal analysis of ERS synthetic aperture radar data. International Journal of Remote Sensing 30: 2619-2635.

- **Conacher, A. J. and Dalrymple, J. B. 1977**. 9 unit land-surface model approach to pedogeomorphic research. Geoderma 18: 1-154.
- Cook, P. G., Walker, G. R. and Jolly, I. D. 1989. Spatial variability of groundwater recharge in a semiarid region. Journal of Hydrology 111: 195-212.
- **Costa-Cabral, M. C. and Burges, S. J. 1994**. Digital elevation model networks (demon) a model of flow over hillslopes for computation of contributing and dispersal areas. Water Resources Research 30: 1681-1692.
- **Coughlan, J. C. and Running, S. W. 1997.** Regional ecosystem simulation: A general model for simulating snow accumulation and melt in mountainous terrain. Landscape Ecology 12: 119-136.
- **Crawford, N. H. and Linsley, R. K. 1966**. Digital simulation in hydrology: Stanford watershed model IV. Dept. of Civil Engineering, Stanford University.
- Creed, I. F., Band, L. E., Foster, N. W., Morrison, I. K., Nicolson, J. A., Semkin, R. S. and Jeffries, D. S. 1996. Regulation of nitrate-N release from temperate forests: A test of the N flushing hypothesis. Water Resources Research 32: 3337-3354.
- **Creed, I. F. and Beall, F. D. 2009**. Distributed topographic indicators for predicting nitrogen export from headwater catchments. Water Resources Research 45: W10407.
- Creed, I. F., Beall, F. D., Clair, T. A., Dillon, P. J. and Hesslein, R. H. 2008a. Predicting export of dissolved organic carbon from forested catchments in glaciated landscapes with shallow soils. Global Biogeochemical Cycles 22: GB4024.
- **Creed, I. F., Sanford, S. E., Beall, F. D., Molot, L. A. and Dillon, P. J. 2003**. Cryptic wetlands: integrating hidden wetlands in regression models of the export of dissolved organic carbon from forested landscapes. Hydrological Processes 17: 3629-3648.
- **Creed, I. F., Sass, G. Z., Wolniewicz, M. B. and Devito, K. J. 2008b**. Incorporating hydrological dynamics into buffer strip design on the sub-humid Boreal Plain of Alberta. Forest Ecology and Management 256: 1984-1994.
- **Creed, I. F., Sass, G. Z., Beall, F. D., Buttle, J. M., Moore, R. D. and Donnelly, M. 2011.** Hydrological principles for conservation of water resources within a changing forested landscape. A State of Knowledge report. Sustainable Forest Management Network, Edmonton, Alberta. 80 pp.
- **Creed, I. F., Tague, C. L., Clark, R., Wolniewicz, M., Zsigovics, G. and Krezek, C. 2002**. Modeling hydrological processes in boreal watersheds: the proof is in the pattern. AGU Fall Meeting 2002, H61B-0775.
- **Croke, B. F. W. and Jakeman, A. J. 2004**. A catchment moisture deficit module for the IHACRES rainfall-runoff model. Environmental Modelling & Software 19: 1-5.
- D'Arcy, P. and Carignan, R. 1997. Influence of catchment topography on water chemistry in southeastern Quebec Shield lakes. Canadian Journal of Fisheries and Aquatic Sciences 54: 2215-2227.
- **da Silva, R. and Avissar, R. 2006**. The hydrometeorology of a deforested region of the Amazon basin. Journal of Hydrometeorology 7: 1028-1042.
- Davis, W. S., and Simon, T.P. 1995. Introduction. In Biological assessment and criteria. Tools for water resource planning and decision making, W. S. Davis and T. P. Simon (eds.). Lewis Publishers, Boca Raton, Florida, 3-6.
- Deng, Y., Wilson, J. P. and Bauer, B. O. 2007. DEM resolution dependencies of terrain attributes across a landscape. International Journal of Geographical Information Science 21: 187-213.

- **Desmet, P. J. J. and Govers, G. 1996**. Comparison of routing algorithms for digital elevation models and their implications for predicting ephemeral gullies. International Journal of Geographical Information Systems 10: 311-331.
- Devito, K. J., Creed, I. F., Gan, T., Mendoza, C., Petrone, R., Silins, U. and Smerdon, B. 2005. A framework for broad-scale classification of hydrological response units on the Boreal Plain: is topography the last thing to consider? Hydrological Processes 19: 1705-1714.
- **Devito, K. J., Creed, I. F., Rothwell, R. L. and Prepas, E. E. 2000**. Landscape controls on phosphorus loading to boreal lakes: implications for the potential impacts of forest harvesting. Canadian Journal of Fisheries and Aquatic Sciences 57: 1977-1984.
- Dhakal, A. S. and Sidle, R.C. 2003. Long-term modeling of landslides for different forest management practices. Earth Surface Processes and Landforms 28: 853– 868.
- Dhakal, A. S. and Sidle, R.C. 2004. Pore water pressure assessment in a forest watershed: Simulations and distributed field measurements related to forest practices. Water Resources Research 40: W02405.
- Díaz, M. F., Bigelow, S. and Armesto J. J. 2007. Alteration of the hydrologic cycle due to forest clearing and its consequences for rainforest succession. Forest Ecology and Management 244: 32-40.
- **Dise, N. B. and Gundersen, P. 2004**. Forest ecosystem responses to atmospheric pollution: linking comparative with experimental studies. Water, Air and Soil Pollution Focus 4: 207-220.
- **Donnelly, M. 2003.** Ecologically-based forest planning and management for aquatic ecosystems in the Duck Mountains, Manitoba. Journal of Environmental Engineering and Science 2: S35-S40.
- Dunne, T. 1978. Field studies of hillslope flow processes. In Hillslope Hydrology, M. J. Kirkby (ed.). Wiley, Chichester.
- **Dunne, T., Moore, T. T., Taylor, C. H. 1975.** Recognition and prediction of runoff-producing zones in humid regions. Hydrological Sciences Bulletin des Sciences Hydrologiques 20: 305-327.
- **Eisenbies, M. H., Aust, W. M., Burger, J. A. and Adams, M. B. 2007**. Forest operations, extreme flooding events, and considerations for hydrological modeling in the Appalachians A review. Forest Ecology and Management 242: 77-98.
- El-Kaddah, D. N. and Carey, A. E. 2004. Water quality modeling of the Cahaba River, Alabama. Environmental Geology 45: 323-338.
- Ellis, C. R., Pomeroy, J. W., Brown, T. and MacDonald, J. 2010. Simulation of snow accumulation and melt in needleleaf forest environments. Hydrology and Earth System Sciences 14: 925-940.
- Entekhabi, D. and Moghaddam, M. 2007. Mapping recharge from space: roadmap to meeting the grand challenge. Hydrogeology Journal 15: 105-116.
- **Evans, J. E., Prepas, E. E., Devito, K. J. and Kotak, B. G. 2000**. Phosphorus dynamics in shallow subsurface waters in an uncut and cut subcatchment of a lake on the Boreal Plain. Canadian Journal of Fisheries and Aquatic Sciences 57: 60-72.
- **Fairfield, J. and Leymarie, P. 1991**. Drainage networks from grid digital elevation models. Water Resources Research 27: 709-717.
- Farr, T. G., Rosen, P. A., Caro, E., Crippen, R., Duren, R., Hensley, S., Kobrick, M., Paller, M., Rodriguez, E., Roth, L., Seal, D., Shaffer, S., Shimada, J., Umland, J., Werner, M., Oskin, M., Burbank, D. and Alsdorf, D. 2007. The shuttle radar topography mission. Reviews of Geophysics 45: RG2004.

- Feller, M. C. 2005. Forest harvesting and streamwater inorganic chemistry in western North America: a review. Journal of the American Water Resources Association 41: 785-811.
- Flaming, G. M. 2005. Global precipitation measurement update. IGARSS 2005: IEEE International Geoscience and Remote Sensing Symposium, Vols 1-8, Proceedings, 79-82.
- Foster, D. Swanson, F., Aber, J., Burke, I., Brokaw, N Tilman, D. and Knapp, A. 2003. The importance of land-use legacies to ecology and conservation. Bioscience 53: 77-88.
- **Fowler, H. J., Blenkinsop, S. and Tebaldi, C. 2007**. Linking climate change modelling to impacts studies: recent advances in downscaling techniques for hydrological modelling. International Journal of Climatology 27: 1547-1578.
- **Franks, S. W., Gineste, P., Beven, K. J. and Merot P. 1998.** On constraining the predictions of a distributed model: The incorporation of fuzzy estimates of saturated areas into the calibration process. Water Resources Research 34: 787-797.
- Francos, A., Bidoglio, G., Galbiati, L., Bouraoui, F., Elorza, F. J., Rekolainen, S., Manni, K. and Granlund, K. 2001. Hydrological and water quality modelling in a medium-sized coastal basin. Physics and Chemistry of the Earth Part B-Hydrology Oceans and Atmosphere 26: 47-52.
- Freer, J., McDonnell, J. J., Beven, K. J., Peters, N. E., Burns, D. A., Hooper, R. P., Aulenbach, B. and Kendall, C. 2002. The role of bedrock topography on subsurface storm flow. Water Resources Research 38: WR000872.
- **Fuller, R. D., Simone, D. M. and Driscoll C. T. 1988**. Forest clearcutting effects on trace metal concentrations: spatial patterns in soil solutions and streams. Water, Air and Soil Pollution 40: 185-195.
- Gao, J. 2009. Bathymetric mapping by means of remote sensing: methods, accuracy and limitations. Progress in Physical Geography 33: 103-116.
- Garcia, E. and Carignan, R. 2000. Mercury concentrations in northern pike (*Esox lucius*) from boreal lakes with logged, burned or undisturbed catchments. Canadian Journal of Fisheries and Aquatic Sciences 57 (Suppl 2): 97-104.
- Garcia-Martinó, A. R., Scatena, F. N., Warner, G. S. and Civco, D. L. 1996. Statistical low flow estimation using GIS analysis in humid montane regions in Puerto Rico. Water Resources Bulletin 32: 1259-1271.
- Gary, H. L. and Watkins, R. K. 1985. Snowpack accumulation before and after thinning a dog-hair stand of Lodgepole pine. U.S. Department of Agriculture Forest Service Resources Note RM-450, Rocky Mountain Forest and Range Experimental Station, Fort Collins, CO.
- Gauthier, S., Vaillancourt, M.A., Kneeshaw, D., Drapeau, P., DeGrandpre, L., Claveau, Y. and Bergeron, Y. (eds.). 2009. Ecosystem management in the Boreal Forest. Les Presses de l'Universite du Quebec, Quebec, Quebec. 13-37.
- Georgakakos, K. P., Seo, D. J., Gupta, H., Schaake, J. and Butts, M. B. 2004. Towards the characterization of streamflow simulation uncertainty through multimodel ensembles. Journal of Hydrology 298: 222-241.
- Gergel, S. E., Turner, M. G. and Kratz, T. K. 1999. Dissolved organic carbon as an indicator of the scale of watershed influence on lakes and rivers. Ecological Applications 9: 1377-1390.
- Glenn, E. P., Huete, A. R., Nagler, P. L., Hirschboeck, K. K. and Brown, P. 2007. Integrating remote sensing and ground methods to estimate evapotranspiration. Critical Reviews in Plant Sciences 26: 139-168.

- Goodrich, D. C., Scott, R., Qi, J., Goff, B., Unkrich, C. L., Moran, M. S., Williams, D., Schaeffer, S., Snyder, K., MacNish, R., Maddock, T., Pool, D., Chehbouni, A., Cooper, D. I., Eichinger, W. E., Shuttleworth, W. J., Kerr, Y., Marsett, R. and Ni, W. 2000. Seasonal estimates of riparian evapotranspiration using remote and in situ measurements. Agricultural and Forest Meteorology 105: 281-309.
- **Govender, M. and Everson, C. S. 2005**. Modelling streamflow from two small South African experimental catchments using the SWAT model. Hydrological Processes 19: 683-692.
- Gravelle, J. A., Ice, G., Link, T. E. and Cook, D. L. 2009. Nutrient concentration dynamics in an inland Pacific Northwest watershed before and after timber harvest. Forest Ecology and Management 257: 1663-1675.
- Gritzner, M. L., Marcus, W. A., Aspinall, R. and Custer, S. G. 2001. Assessing landslide potential using GIS, soil wetness modeling and topographic attributes, Payette River, Idaho. Geomorphology 37: 149-165.
- Guerschman, J. P., Van Dijk, A. I. J. M., Mattersdorf, G., Beringer, J., Hutley, L. B., Leuning, R., Pipunic, R. C. and Sherman, B. S. 2009. Scaling of potential evapotranspiration with MODIS data reproduces flux observations and catchment water balance observations across Australia. Journal of Hydrology 369: 107-119.
- **Guillemette, F., Plamondon, A. P., Prevost, M. and Levesque, D. 2005**. Rainfall generated stormflow response to clearcutting a boreal forest: peak flow comparison with 50 world-wide basin studies. Journal of Hydrology 302: 137-153.
- **Güntner, A., Seibert, J. and Uhlenbrook, S. 2004**. Modeling spatial patterns of saturated areas: An evaluation of different terrain indices. Water Resources Research 40: W05114.
- Hall, D. K. 1996. Remote sensing applications to hydrology: Imaging radar. Hydrological Sciences Journal Journal Des Sciences Hydrologiques 41: 609-624.
- Hall, G. F. and Olson, C. G. 1991. Predicting variability of soils from landscape models. In Spatial variabilities of soils and landforms, M. J. Manusbach and L. P. Wilding (eds.). Soil Science Society of America Special Publication, No. 28, Madison, WI: 9-24.
- Hamilton, L. S. 1985. Overcoming myths about soil and water impacts of tropical forest land uses. In Soil Erosion and Conservation, S. A. El-Swaify, W. C. Moldenhauer and A. Lo (eds.), Soil Conservation Society of America, Ankey, IA.
- Harr, R. D. 1982. Fog drip in the Bull Run Municipal Watershed, Oregon. Water Resources Bulletin 18: 785-789.
- He, H. M., Zhou, J. and Zhang, W. C. 2008. Modelling the impacts of environmental changes on hydrological regimes in the Hei River Watershed, China. Global and Planetary Change 61: 175-193.
- Henderson-Sellers, A. and Pitman, A. J. 2002. Comments on "Suppressing impacts of the Amazonian deforestation by the global circulation change" Bulletin of the American Meteorological Society 83: 1657-1661.
- Hetherington, E. D. 1987. The importance of forests in the hydrological regime. In Canadian aquatic resources, M. C. Healy and R. R. Wallace (eds.). Department of Fisheries and Oceans, Ottawa: 179-211.
- Hewlett, J. D. and Hibbert A. R. 1967. Factors affecting the response of small watersheds to precipitation in humid areas. In International symposium on forest hydrology, W. E. Sopper and H. W. Lull (eds.). 275-290.
- Hewlett, J. and Nutter W. L. 1970. The varying source area of streamflow from upland basins. In Proceedings of the symposium on interdisciplinary aspects of watershed management. Montana State University, Bozeman, August 3-6.
- Hibbert, A. R. 1967. Forest treatment effects on water yield. In Proceedings of International Symposium on Forest Hydrology, Pennsylvania State University, 1965, W. E. Sopper and H. W. Lull (eds.). Pergamon Press, New York.

- Hjerdt, K. N., McDonnell, J. J., Seibert, J. and Rodhe, A. 2004. A new topographic index to quantify downslope controls on local drainage. Water Resources Research 40: W05602.
- Hong, Y., Gochis, D., Cheng, J. T. Hsu, K. L. and Sorooshian S. 2007. Evaluation of PERSIANN-CCS rainfall measurement using the NAME Event Rain Gauge Network. Journal of Hydrometeorology 8: 469-482.
- Hornbeck, J. W., Martin, C. W. and Eagar, C. 1997. Summary of water yield experiments at Hubbard Brook Experimental Forest, New Hampshire. Canadian Journal of Forest Research 27: 2043-2052.
- Hornbeck, J. W., Adams, M. B., Corbett, E. S., Verry, E. S. and Lynch, J. A. 1993. Long-term impacts of forest treatments on water yield a summary for northeastern USA. Journal of Hydrology 150: 323-344.
- Hornberger, G. M., Bencala, K. E. and McKnight, D. M. 1994. Hydrological controls on dissolved organic-carbon during snowmelt in the Snake River near Montezuma, Colorado. Biogeochemistry 25: 147-165.
- Horton, R. E. 1945. Erosional development of streams and their drainage basins hydrophysical approach to quantitative morphology. Geological Society of America Bulletin 56: 275-370.
- Hubbart, J. A., Link, T. E., Gravelle, J. A., Elliot, W. J. 2007. Timber harvest impacts on water yield in the continental/maritime hydroclimatic region of the United States. Forest Science 53: 169-180.
- Hunter Jr., M.L. 1993. Natural fire regimes as spatial models for managing boreal forests. Biological Conservation 65: 115-120.
- Hutchinson, M. F. 1989. A new procedure for gridding elevation and stream line data with automatic removal of spurious pits. Journal of Hydrology 106: 211-232.
- Hwang, T., Band, L. and Hales, T. C. 2009. Ecosystem processes at the watershed scale: Extending optimality theory from plot to catchment. Water Resources Research 45: W11425.
- Inamdar, S. P. and Mitchell, M. J. 2006. Hydrological and topographic controls on storm-event exports of dissolved organic carbon (DOC) and nitrate across catchment scales. Water Resources Research 42: W03421.
- Istanbulluoglu, E., Tarboton, D. G., Pack, R. T. and Luce, C. 2003. A sediment transport model for incision of gullies on steep topography. Water Resources Research 39: 1103.
- Istanbulluoglu, E., Tarboton, D. G., Pack, R. T. and Luce, C. H. 2004. Modeling of the interactions between forest vegetation, disturbances, and sediment yields. Journal of Geophysical Research-Earth Surface 109: F01009.
- Jackson, T. J. 2002. Remote sensing of soil moisture: implications for groundwater recharge. Hydrogeology Journal 10: 40-51.
- Jackson, T. J., Schmugge, J. and Engman, E. T. 1996. Remote sensing applications to hydrology: Soil moisture. Hydrological Sciences Journal-Journal Des Sciences Hydrologiques 41: 517-530.
- Jain, S. K. and Sudheer, K. P. 2008. Fitting of hydrological models: A close look at the Nash-Sutcliffe index. Journal of Hydrological Engineering 13: 981-986.
- Jakeman, A. J. and Hornberger, G. M. 1993. How much complexity is warranted in a rainfall-runoff model. Water Resources Research 29: 2637-2649.
- Jakeman, A. J., Letcher, R. A. and Norton, J. P. 2006. Ten iterative steps in development and evaluation of environmental models. Environmental Modelling & Software 21: 602-614.
- Jencso, K. G., McGlynn, B. L., Gooseff, M. N., Wondzell, S. M., Bencala, K. E. and Marshall, L. A. 2009. Hydrological connectivity between landscapes and streams: Transferring reach-and plot-scale understanding to the catchment scale. Water Resources Research 45: W04438.

- Jewett, K., Daugharty, D., Krause, H. H. and Arp P. A. 1995. Watershed responses to clear cutting effects on soil solutions and stream water discharge in central New Brunswick. Canadian Journal of Soil Science 75: 475-490.
- Jha, M. K., Chowdhury, A., Chowdary, V. M. and Peiffer, S. 2007. Groundwater management and development by integrated remote sensing and geographic information systems: prospects and constraints. Water Resources Management 21: 427-467.
- Jones, J. A. and Grant, G. E. 1996. Peak flow responses to clear-cutting and roads in small and large basins, western Cascades, Oregon. Water Resources Research 32: 959-974.
- Jones, J. A., Achterman, G. L., Augustine, L. A., Creed, I. F., Ffolliott, P. F., MacDonald, L. and Wemple, B. C. 2009. Hydrological effects of a changing forested landscape-challenges for the hydrological sciences. Hydrological Processes 23: 2699-2704.
- Jost, G., Moore, R. D., Weiler, M., Gluns, D. and Alila, Y. 2009. Use of distributed snow measurements to test and improve a snowmelt model for predicting the effect of forest clear-cutting. Journal of Hydrology 376: 94-106.
- Kaheil, Y. H. and Creed, I. F. 2009. Detecting and downscaling wet areas on boreal landscapes. Ieee Geoscience and Remote Sensing Letters 6: 179-183.
- Kaheil, Y. H., Rosero, E., Gill, M. K., McKee, M. and Bastidas, L. A. 2008. Downscaling and forecasting of evapotranspiration using a synthetic model of wavelets and support vector machines. IEEE Transactions on Geoscience and Remote Sensing 46: 2692-2707.
- Kalma, J. D., McVicar, T. R. and McCabe, M. F. 2008. Estimating Land Surface Evaporation: A Review of Methods Using Remotely Sensed Surface Temperature Data. Surveys in Geophysics 29: 421-469.
- Kasischke, E. S., Melack, J. M. and Dobson, M. C. 1997. The use of imaging radars for ecological applications A review. Remote Sensing of Environment 59: 141-156.
- **Kimmins, J.P. 2004.** Forest Ecology: a foundation for sustainable forest management and environmental ethics in forestry. Prentice Hall, Upper Saddle River, New Jersey. 611 pp.
- Kirchner, J. W. 2006. Getting the right answers for the right reasons: Linking measurements, analyses, and models to advance the science of hydrology. Water Resources Research 42: W03S04.
- Kite, G. W. and Pietroniro, A. 1996. Remote sensing applications in hydrological modelling. Hydrological Sciences Journal-Journal Des Sciences Hydrologiques 41: 563-591.
- Kittel, T. G. F., Rosenbloom, N. A., Painter, T. H., Schimel, D. S., Melillo, J. M., Pan, Y. D., Kicklighter, D. W., McGuire, A. D., Neilson, R. P., Chaney, J., Ojima, D. S., McKeown, R., Parton, W. J., Pulliam, W. M., Prentice, I. C., Haxeltine, A., Running, S. W., Pierce, L. L., Nemani, R. R., Hunt, E. R., Smith, T. M., Rizzo, B. and Woodward, F. I. 1995. The VEMAP integrated database for modelling United States ecosystem/vegetation sensitivity to climate change. Journal of Biogeography 22: 857-862.
- Kittredge, J. 1948. Forest Influences. McGraw Hill, New York.
- Klein, A. G., Hall, D. K. and G. A. Riggs. 1998. Improving snow cover mapping in forests through the use of a canopy reflectance model. Hydrological Process 12: 1723-1744.
- Klemeš, V. 1986a. Dilettantism in hydrology: transition or destiny? Water Resources Research 22 (9): 177S-188S.
- Klemeš, V. 1986b. Operational testing of hydrological simulation-models. Hydrological Sciences Journal-Journal Des Sciences Hydrologiques 31: 13-24.

- Klenk, N. L., Bull, G. Q. and MacLellan, J. I. 2009. The "emulation of natural disturbance" (END) management approach in Canadian forestry: a critical evaluation. Forestry Chronicle 85: 440-445.
- Krezek, C. C., Buttle, J. M., Beall, F. D., Moore, R. D., Creed, I. F., Sibley, P. K., Silins, U., Devito, K. J. and Mendoza, C. A. 2008. HydroEcological Landscapes and Processes Project: A National-scale Forest Hydrology Initiative. Streamline Watershed Management Bulletin 12: 33-38.
- **Krougly, Z. L., Creed, I. F. and Stanford, D. A. 2009**. A stochastic model for generating disturbance patterns within landscapes. Computers & Geosciences 35: 1451-1459.
- Kuczera, G., Raper, G. P., Brah, N. S. and Jayasuriya, M. D. 1993. Modeling yield changes after strip thinning in a mountain ash catchment an exercise in catchment model validation. Journal of Hydrology 150: 433-457.
- Kustas, W. P. and Norman, J. M. 1996. Use of remote sensing for evapotranspiration monitoring over land surfaces. Hydrological Sciences Journal-Journal Des Sciences Hydrologiques 41: 495-516.
- Landres, P. B., Morgan, P. and Swanson, F. J. 1999. Overview of the use of natural variability concepts in managing ecological systems. Ecological Applications 9: 1179-1188.
- Lane, S. N., Brookes, C. J., Kirkby, A. J. and Holden, J. 2004. A network-indexbased version of TOPMODEL for use with high-resolution digital topographic data. Hydrological Processes 18: 191-201.
- Lang, M. W. and Kasischke, E. S. 2008. Using C-band synthetic aperture radar data to monitor forested wetland hydrology in Maryland's coastal plain, USA. IEEE Transactions on Geoscience and Remote Sensing 46: 535-546.
- Lang, M. W. and McCarty, G. W. 2009. LiDAR intensity for improved detection of inundation below the forest canopy. Wetlands 29: 1166-1178.
- Laudon, H., Sjoblom, V., Buffam, I., Seibert, J. and Morth, M. 2007. The role of catchment scale and landscape characteristics for runoff generation of boreal streams. Journal of Hydrology 344: 198-209.
- Lawton, R. O., Nair, U. S., Pielke Jr., R. A. and Welch, R. M. 2001. Climatic impact of tropical lowland deforestation on nearby montane cloud forests. Science 294: 584-587.
- Lehmann, P., Hinz, C., McGrath, G., Tromp-van Meerveld, H. J. and McDonnell, J. J. 2007. Rainfall threshold for hillslope outflow: an emergent property of flow pathway connectivity. Hydrology and Earth System Sciences 11: 1047-1063.
- Lemmetyinen, J., Derksen, C., Pulliainen, J., Strapp, W., Toose, P., Walker, A., Tauriainen, S., Plhlflyckt, J., Karna, J.-P. and Hallikainen, M. T. 2009. A Comparison of Airborne Microwave Brightness Temperatures and Snowpack Properties Across the Boreal Forests of Finland and Western Canada. IEEE Transactions on Geoscience and Remote Sensing 47: 965-978.
- Levia, D. F., Carlyle-Moses, D. E. and Tanaka T. (Eds.). 2011. Forest Hydrology and Biogeochemistry: Synthesis of Past Research and Future Directions (pp.740). Ecological Studies Series, No. 216, Springer-Verlag, Heidelberg, Germany.
- Li, X., Li, X. W., Li, Z. Y., Ma, M. G., Wang, J., Xiao, Q., Liu, Q., Che, T., Chen, E. X., Yan, G. J., Hu, Z. Y., Zhang, L. X., Chu, R. Z., Su, P. X., Liu, Q. H., Liu, S. M., Wang, J. D., Niu, Z., Chen, Y., Jin, R., Wang, W. Z., Ran, Y. H., Xin, X. Z. and Ren, H. Z. 2009a. Watershed Allied Telemetry Experimental Research. Journal of Geophysical Research-Atmospheres 114: D22103.
- Li, Z. L., Tang, R. L., Wan, Z. M., Bi, Y. Y., Zhou, C. H., Tang, B. H., Yan, G. J. and Zhang, X. Y. 2009b. A Review of Current Methodologies for Regional Evapotranspiration Estimation from Remotely Sensed Data. Sensors 9: 3801-3853.

- Likens, G. E., Bormann, F. H., Johnson, N. M. 1969. Nitrification: importance to nutrient losses from a cutover forested ecosystem. Science 163: 1205-1206.
- Lindgren, G. A., Wrede, S., Seibert, J. and Wallin, M. 2007. Nitrogen source apportionment modeling and the effect of land-use class related runoff contributions. Nordic Hydrology 38: 317-331.
- Lindo, Z. and Visser, S. 2003. Microbial biomass, nitrogen and phosphorus mineralization, and mesofauna in boreal conifer and deciduous forest floors following partial and clear-cut harvesting. Canadian Journal of Forest Research 33: 1610-1620.
- Lindsay, J. B. 2003. A physically based model for calculating contributing area on hillslopes and along valley bottoms. Water Resources Research 39: 1332.
- Lindsay, J. B. 2005. The Terrain Analysis System: a tool for hydro-geomorphic applications. Hydrological Processes 19: 1123-1130.
- **Lindsay, J. B. 2006**. Sensitivity of channel mapping techniques to uncertainty in digital elevation data. International Journal of Geographical Information Science 20: 669-692.
- Lindsay, J. B. and Creed, I. F. 2005. Removal of artifact depressions from digital elevation models: towards a minimum impact approach. Hydrological Processes 19: 3113-3126.
- Lindsay, J. B. and Creed, I. F. 2006. Distinguishing actual and artefact depressions in digital elevation data. Computers & Geosciences 32: 1192-1204.
- Lindsay, J. B., Creed, I. F. and Beall, F. D. 2004. Drainage basin morphometrics for depressional landscapes. Water Resources Research 40: W09307.
- Liu, J., Chen, J. M. and Cihlar, J. 2003. Mapping evapotranspiration based on remote sensing: An application to Canada's landmass. Water Resources Research 39: 1189.
- Loftis, J. C., MacDonald, L. H., Streett, S., Iyer, H. K. and Bunte, K. 2001. Detecting cumulative watershed effects: the statistical power of pairing. Journal of Hydrology 251: 49-64.
- Loukas, A., Vasiliades, L. and Dalezios, N. R. 2002. Climatic impacts on the runoff generation processes in British Columbia, Canada. Hydrology and Earth System Sciences 6: 211-227.
- Lu, J.B., Sun, G., McNulty, S. G. and Amatya, D. M. 2003. Modeling actual evapotranspiration from forested watersheds across the Southeastern United States. Journal of the American Water Resources Association 39: 887-896.
- Lu, Z. and Kwoun, O. I. 2008. Radarsat-1 and ERS InSAR analysis over southeastern coastal Louisiana: Implications for mapping water-level changes beneath swamp forests. IEEE Transactions on Geoscience and Remote Sensing 46: 2167-2184.
- Lundberg, A., Nakai, Y., Thunehed, H. and Halldin, S. 2004. Snow accumulation in forests from ground and remote-sensing data. Hydrological Processes 18: 1941-1955.
- Lunetta, R. S. and Ballogh, M. E. 1999. Application of multi-temporal Landsat 5 TM imagery for wetland identification. Photogrammetric Engineering and Remote Sensing 65: 1303-1310.
- Ma, X., Xu, J. C. and van Noordwijk, M. 2010. Sensitivity of streamflow from a Himalayan catchment to plausible changes in land cover and climate. Hydrological Processes 24: 1379-1390.
- Mackay, D. S., Ahl, D. E., Ewers, B. E., Gower, S. T., Burrows, S. N., Samanta, S. and Davis, K. J. 2002. Effects of aggregated classifications of forest composition on estimates of evapotranspiration in a northern Wisconsin forest. Global Change Biology 8: 1253-1265.

- MacMillan, R. A., Pettapiece, W. W., Nolan, S. C. and Goddard, T. W. 2000. A generic procedure for automatically segmenting landforms into landform elements using DEMs, heuristic rules and fuzzy logic. Fuzzy Sets and Systems 113: 81-109.
- Madsen, H. 2000. Automatic calibration of a conceptual rainfall-runoff model using multiple objectives. Journal of Hydrology 235: 276-288.
- Mallik, A. and Teichert, S. 2009. Effects of forest Management on water resources in Canada: a research review. NCASI Technical Bulletin No. 969. December 2009.
- Mark, D. M. and Goodchild, M. F. 1982. Topologic model for drainage networks with lakes. Water Resources Research 18: 275-280.
- Marland, G., Pielke Jr., R. A., Apps, M., Avissar, R., Betts, R. A., Davis, K. J., Frumhoff, P. C., Jackson, S. T., Joyce, L. A., Kauppi, P., Katzenberger, J., MacDicken, K. G., Nelson, R. P., Niles, J. O., Niyogi, D. S., Norby, R. J., Pena, N., Sampson, N. and Xue, Y. 2003. The climatic impacts of land surface change and carbon management, and the implications for climate-change mitigation policy. Climate Policy 3: 149-157.
- Martin, J. G., Bolstad, P. V., Ryu, S. R. and Chen, J. Q. 2009. Modeling soil respiration based on carbon, nitrogen, and root mass across diverse Great Lake forests. Agricultural and Forest Meteorology 149: 1722-1729.
- Martz, L. W. and Garbrecht, J. 1998. The treatment of flat areas and depressions in automated drainage analysis of raster digital elevation models. Hydrological Processes 12: 843-855.
- Mascarelli, A. L. 2009. A sleeping giant? Nature Reports 3: 46-49.
- McClain, M. E., Boyer, E. W., Dent, C. L., Gergel, S. E., Grimm, N. B., Groffman, P. M., Hart, S. C., Harvey, J. W., Johnston, C. A., Mayorga, E., McDowell, W. H. and Pinay, G. 2003. Biogeochemical hot spots and hot moments at the interface of terrestrial and aquatic ecosystems. Ecosystems 6: 301-312.
- McCuen, R. H., Knight, Z. and Cutter, A. G. 2006. Evaluation of the Nash-Sutcliffe efficiency index. Journal of Hydrological Engineering 11: 597-602.
- McDonnell, J. J., Sivapalan, M., Vache, K., Dunn, S., Grant, G., Haggerty, R., Hinz, C., Hooper, R., Kirchner, J., Roderick, M. L., Selker, J. and Weiler, M. 2007. Moving beyond heterogeneity and process complexity: A new vision for watershed hydrology. Water Resources Research 43: W07301.
- McGlynn, B., McDonnell, J., Stewart, M. and Seibert, J. 2003. On the relationships between catchment scale and streamwater mean residence time. Hydrological Processes 17: 175-181.
- McGlynn, B. L. and Seibert, J. 2003. Distributed assessment of contributing area and riparian buffering along stream networks. Water Resources Research 39: 1082.
- McGuire, K. J. and McDonnell, J. J. 2006. A review and evaluation of catchment transit time modeling. Journal of Hydrology 330: 543-563.
- McGuire, K. J., McDonnell, J. J., Weiler, M., Kendall, C., McGlynn, B. L., Welker, J. M. and Seibert, J. 2005. The role of topography on catchment-scale water residence time. Water Resources Research 41.
- McHale, M. R., Burns, D. A., Lawrence, G. B., Murdoch, P. S. 2007. Factors controlling soil water and stream water aluminum concentrations after a clearcut in a forested watershed with calcium-poor soils. Biogeochemistry 84: 311-331.
- McMaster, K. J. 2002. Effects of digital elevation model resolution on derived stream network positions. Water Resources Research 38: W04002.

- McRae, DJ, Duchesne, L.C., Freedman, B, Lynham, T.J. and Woodley, S. 2001. Comparisons between wildfire and forest harvesting and their implications in forest management. Environ. Rev. 9: 223-260.
- Megahan, W. F. and King, J. G. 2001. Erosion, sedimentation, and cumulative effects in the northern Rocky Mountains, Proceedings of The Society of American Foresters pp. 58-63.
- Meijerink, A. M. J. 1996. Remote sensing applications to hydrology: Groundwater. Hydrological Sciences Journal-Journal Des Sciences Hydrologiques 41: 549-561.
- Meijerink, A. M. J. 2000. Groundwater. In remote sensing in hydrology and water management, G. Schultz (ed.). Berlin: Springer-Verlag. 305-325.
- Merz, R., Parajka, J. and Bloschl, G. 2009. Scale effects in conceptual hydrological modeling. Water Resources Research 45: W09405.
- Milne, G. 1935. Some suggested units of classification and mapping, particularly for East African soils. Soil Research 3: 181-198.
- Min, Q. L. and Lin, B. 2006. Remote sensing of evapotranspiration and carbon uptake at Harvard Forest. Remote Sensing of Environment 100: 379-387.
- Molotch, N. P. 2009. Reconstructing snow water equivalent in the Rio Grande headwaters using remotely sensed snow cover data and a spatially distributed snowmelt model. Hydrological Processes 23: 1076-1089.
- Montgomery, D. R. and Buffington, J. M. 1997. Channel-reach morphology in mountain drainage basins. GSA Bulletin 109: 596-611.
- Montgomery, D. R. and Dietrich, W. E. 1992. Channel initiation and the problem of landscape scale. Science 255: 826-830.
- Montgomery, D. R., Schmidt, K. M., Greenberg, H. M. and Dietrich, W.E. 2000. Forest clearing and regional landsliding. Geology 28: 311–314.
- Moore, I. D., Grayson, R. B. and Ladson, A. R. 1991. Digital terrain modeling a review of hydrological, geomorphological, and biological applications. Hydrological Processes 5: 3-30.
- Moore, R. D. and Wondzell, S. M. 2005. Physical hydrology and the effects of forest harvesting in the Pacific Northwest: A review. Journal of the American Water Resources Association 41: 763-784.
- Moore, R. D., Spittlehouse, D. L. and Story, A. 2005. Riparian microclimate and stream temperature response to forest harvesting: a review. Journal of the American Water Resources Association 41: 813-834.
- Moran, M. S., Clarke, T. R., Inoue, Y. and Vidal, A. 1994. Estimating crop water-deficit using the relation between surface-air temperature and spectral vegetation index. Remote Sensing of Environment 49: 246-263.
- Moran, M. S., Peters-Lidard, C. D., Watts, J. M. and McElroy, S. 2004. Estimating soil moisture at the watershed scale with satellite-based radar and land surface models. Canadian Journal of Remote Sensing 30: 805-826.
- Mu, Q., Heinsch, F. A., Zhao, M. and Running, S. W. 2007. Development of a global evapotranspiration algorithm based on MODIS and global meteorology data. Remote Sensing of Environment 111: 519-536.
- Murphy, P. N. C., Ogilvie, J., Castonguay, M., Zhang, C. F., Meng, F. R. and Arp, P. A. 2008. Improving forest operations planning through high-resolution flow-channel and wet-areas mapping. Forestry Chronicle 84: 568-574.
- Murphy, P. N. C., Ogilvie, J., Connor, K. and Arp, P. A. 2007. Mapping wetlands: A comparison of two different approaches for New Brunswick, Canada. Wetlands 27: 846-854.

- Murray, C. D. and Buttle, J. M. 2003. Impacts of clearcut harvesting on snow accumulation and melt in a northern hardwood forest. Journal of Hydrology 271: 197-212.
- Naiman, R.J., Decamps, H. and McClain, M.E. 2005. Riparia ecology, conservation and management of streamside communities. Elsevier Academic Press, Burlington, MA.
- Nash, J. E. and Sutcliffe, J. V. 1970. River flow forecasting through conceptual models part I-A discussion of principles. Journal of Hydrology 10: 282-290.
- National Research Council. 1992. Restoration of aquatic systems: science, technology and public policy. National Academy Press, Washington, D.C.
- National Research Council of the National Academies. 2008. Hydrologic Effects of a Changing Forest Landscape. The National Academies Press, Washington, D.C.
- Natural Resources Canada. 2007. Canadian Digital Elevation Data, Level 1, Product Specifications, Government of Canada, Natural Resources Canada, Centre for Topographic Information.
- Nemani, R. Pierce, L., Running, S. and Band, L. E. 1993. Forest ecosystem processes at the watershed scale sensitivity to remotely-sensed leaf-area index estimates. International Journal of Remote Sensing 14: 2519-2534.
- Nour, M. H., Smith, D. W., El-Din, M. G. and Prepas, E. E. 2006. Artificial neural networks and time series modelling of TP concentration in boreal streams: a comparative approach. Journal of Environmental Engineering and Science 5: S39-S52.
- O'Connor, E. M., Dillon, P. J., Molot, L. A. and Creed, I. F. 2009. Modeling dissolved organic carbon mass balances for lakes of the Muskoka River Watershed. Hydrology Research 40: 273-290.
- **O'Callaghan, J. F. and Mark, D. M. 1984**. The extraction of drainage networks from digital elevation data. Computer Vision Graphics and Image Processing 28: 323-344.
- Oda, T., Asano, Y., Suzuki, M. 2009. Transit time evaluation using a chloride concentration input step shift after forest cutting in a Japanese headwater catchment. Hydrological Processes 23: 2705-2713.
- The Ontario Forest Management Guide for Natural Disturbance Emulation (www.mnr.gov.on.ca/en/Business/Forests/Publication/MNR_E000509P.html).
- **Ontario Ministry of Natural Resources. 2001.** Forest management guide for natural disturbance pattern emulation. Vers. 3.1. Queens Printer for Ontario. 211 pp.
- **Oudin, L., Weisse, A., Loumagne, C. and Le Hegarat-Mascle, S. 2003**. Assimilation of soil moisture into hydrological models for flood forecasting: a variational approach. Canadian Journal of Remote Sensing 29: 679-686.
- Paik, K. 2008. Global search algorithm for nondispersive flow path extraction. Journal of Geophysical Research-Earth Surface 113: F04001.
- Peck, A. J. and Williamson, D. R. 1987. Effects of forest clearing on groundwater. Journal of Hydrology 94: 47-65.
- Pereira, H. C. 1973. Land use and water resources in temperate and tropical climates. Cambridge University Press, London.
- Perera, A. H. and Buse, L. J. 2004. Emulating natural disturbance in forest management: an overview. In Emulating natural forest landscape disturbances: concepts and applications, A. H. Perera, L. J. Buse, and M.G. Weber (eds.). Columbia University Press, New York. 3-7.

- Peters, D. L., Buttle, J. M., Taylor, C. H. and LaZerte, B. D. 1995. Runoff production in a forested, shallow soil Canadian Shield basin. Water Resources Research 31: 1291-1304.
- **Petty, G. W. 1995**. The status of satellite-based rainfall estimation over land. Remote Sensing of Environment 51: 125-137.
- **Petty, G. W. and Krajewski, W. F. 1996**. Satellite estimation of precipitation over land. Hydrological Sciences Journal-Journal Des Sciences Hydrologiques 41: 433-451.
- **Peucker, T. K. Douglas, D. H. 1975**. Detection of surface-specific points by local parallel processing of discrete terrain elevation data. Computer Vision Graphics and Image Processing 4: 375-387.
- Pike, R. G., Spittlehouse, D. L., Bennett, K. E., Egginton, V. N., Tschaplinski, P. G., Murdock, T. Q. and Werner, A. T. 2008. Climate change and watershed hydrology: Part II – Hydrological implications for British Columbia. Streamline Watershed Management Bulletin 11: 8-13.
- **Planchon, O. and Darboux, F. 2002**. A fast, simple and versatile algorithm to fill the depressions of digital elevation models. Catena 46: 159-176.
- Poff, N. L., Allan, J. D., Bain, M. B., Karr, J. R., Prestegaard, K. L., Richter, B. D., Sparks, R. E. and Stromberg, J. C. 1997. The natural flow regime. Bioscience 47: 769-784.
- **Pomeroy, J. W., Parviainen, J., Hedstrom, N. R. and Gray, D. M. 1998**. Coupled modelling of forest snow interception and sublimation. Hydrological Processes 12: 2317-2337.
- Pomeroy, J. W., Gray, D. M., Hedstrom, N. R. and Janowicz, J. R. 2002. Prediction of seasonal snow accumulation in cold climate forests. Hydrological Processes 16: 3543-3558.
- **Povari, P., Verta, M., Munthe, J. and Haapanen, M. 2003**. Forestry practices increase mercury and methyl mercury output from boreal forest catchments. Environmental Science and Technology 37: 2389-2393.
- **Prevost, M., Plamondon, A. P. and Belleau, P. 1999**. Effects of drainage of a forested peatland on water quality and quantity. Journal of Hydrology 214: 130-143.
- **Puech, C. and Gineste, P. 2003**. Radar imagery and saturated areas: decreasing model equifinality. Canadian Journal of Remote Sensing 29: 729-733.
- Putz, G., Burke, J. M., Smith, D. W., Chanasyk, D. S., Prepas, E. E. and Mapfumo, E. 2003. Modelling the effects of boreal forest landscape management upon streamflow and water quality: Basic concepts and considerations. Journal of Environmental Engineering and Science 2: S87-S101.
- Quinn, P., Beven, K., Chevallier, P. and Planchon, O. 1991. The prediction of hillslope flow paths for distributed hydrological modeling using digital terrain models. Hydrological Processes 5: 59-79.
- Quinn, P. F., Beven, K. J. and Lamb, R. 1995. The ln(a/tan-beta) index how to calculate it and how to use it within the topmodel framework. Hydrological Processes 9: 161-182.
- Ramillien, G., Famiglietti, J. S. and Wahr, J. 2008. Detection of Continental Hydrology and Glaciology Signals from GRACE: A Review. Surveys in Geophysics 29: 361-374.
- **Rango, A. 1996**. Spaceborne remote sensing for snow hydrology applications. Hydrological Sciences Journal-Journal Des Sciences Hydrologiques 41: 477-494.
- **Refsgaard, J. C. and Knudsen, J. 1996**. Operational validation and intercomparison of different types of hydrological models. Water Resources Research 32: 2189-2202.

- **Remmel, T. K., Todd, K. W. and Buttle, J. 2008**. A comparison of existing surficial hydrological data layers in a low-relief forested Ontario landscape with those derived from a LiDAR DEM. Forestry Chronicle 84: 850-865.
- **Reuss, J. O., Stottlemeyer, R. and Troendle, C. A. 1997**. Effect of clear cutting on nutrient fluxes in a subalpine forest at Fraser, Colorado. Hydrology and Earth System Sciences 1: 333-344.
- Reutebuch, S. E., McGaughey, R. J., Andersen, H. E. and Carson, W. W. 2003. Accuracy of a high-resolution lidar terrain model under a conifer forest canopy. Canadian Journal of Remote Sensing 29: 527-535.
- **Reynoldson, T. B., Norris, R. H., Resh, V. H., Day, K. E. and Rosenberg, D. M. 1997**. The reference condition: A comparison of multimetric and multivariate approaches to assess water-quality impairment using benthic macroinvertebrates. Journal of the North American Benthological Society 16: 833-852.
- Richardson, M. C., Fortin, M. J. and Branfireun, B. A. 2009. Hydrogeomorphic edge detection and delineation of landscape functional units from lidar digital elevation models. Water Resources Research 45: W10441.
- Richter, B. D., Baumgartner, J. V., Wigington, R. and Braun, D. P. 1997. How much water does a river need? Freshwater Biology 37: 231-249.
- Ritchie, J. C. 1996. Remote sensing applications to hydrology: Airborne laser altimeters. Hydrological Sciences Journal-Journal Des Sciences Hydrologiques 41: 625-636.
- **Riveros-Iregui, D. A. and McGlynn, B. L. 2009**. Landscape structure control on soil CO₂ efflux variability in complex terrain: Scaling from point observations to watershed scale fluxes. Journal of Geophysical Research-Biogeosciences 114: G02010.
- Roberts D. W., Dowling, T. I. and Walker J. 1997. FLAG: A fuzzy landscape analysis GIS method for dryland salinity assessment, Tech. Rep. 8/97, CSIRO Land and Water, Canberra.
- Robinson, D. A., Binley, A., Crook, N., Day-Lewis, F. D., Ferre, T. P. A., Grauch, V. J. S., Knight, R., Knoll, M., Lakshmi, V., Miller, R., Nyquist, J., Pellerin, L., Singha, K. and Slater, L. 2008. Advancing process-based watershed hydrological research using near-surface geophysics: a vision for, and review of, electrical and magnetic geophysical methods. Hydrological Processes 22: 3604-3635.
- Rosbjerg, D. and Madsen, H. 2005. Concepts of hydrologic modeling. In Encyclopedia of Hydrological Sciences, M. G. Anderson (ed.). Wiley: Chichester.
- **Rosenqvist, A., Shimada, M., Ito, N. and Watanabe, M. 2007**. ALOS PALSAR: A Pathfinder mission for global-scale monitoring of the environment. IEEE Transactions on Geoscience and Remote Sensing 45: 3307-3316.
- Roth, B. E., Slatton, K. C. and Cohen, M. J. 2007. On the potential for high-resolution lidar to improve rainfall interception estimates in forest ecosystems. Frontiers in Ecology and the Environment 5: 421-428.
- Running, S. W., Nemani, R. R., Peterson, D. L., Band, L. E., Potts, D. F., Pierce, L. L. and Spanner, M. A. 1989. Mapping regional forest evapotranspiration and photosynthesis by coupling satellite data with ecosystem simulation. Ecology 70: 1090-1101.
- Sahin, V. and Hall, M. J. 1996. The effects of afforestation and deforestation on water yields. Journal of Hydrology 178: 293-309.
- Salati, E., DaU'Olio, A., Matsui, E., Gat, J. R. 1979. Recycling of water in the Amazon Basin: An isotopic study. Water Resources Research 15: 1250-1258.

- Sanford, S. E., Creed, I. F., Tague, C. L., Beall, F. D. and Buttle, J. M. 2007. Scale-dependence of natural variability of flow regimes in a forested landscape. Water Resources Research 43: W08414.
- Sass, G. Z. and Creed, I. F. 2008. Characterizing hydrodynamics on boreal landscapes using archived synthetic aperture radar imagery. Hydrological Processes 22: 1687-1699.
- Sass, G. Z., Creed, I. F. and Devito, K. J. 2008. Spatial heterogeneity in trophic status of shallow lakes on the Boreal Plain: Influence of hydrological setting. Water Resources Research 44: W08444.
- Sayama, T. and McDonnell, J. J. 2009. A new time-space accounting scheme to predict stream water residence time and hydrograph source components at the watershed scale. Water Resources Research 45: W07401.
- Schultz, G. A. 1996. Remote sensing applications to hydrology: Runoff. Hydrological Sciences Journal-Journal Des Sciences Hydrologiques 41: 453-475.
- Schofield, N. J. 1996. Forest management impacts on water values. Recent Reseach Developments in Hydrology 1: 1-20.
- Schumm, S. A. 1956. The role of creep and rainwash on the retreat of badland slopes. American Journal of Science 254: 693-706.
- Scott, D. F. 1997. The contrasting effects of wildfire and clearfelling on the hydrology of a small catchment. Hydrological Processes 11: 543-555.
- Seibert, J. and McDonnell, J. J. 2010. Land-cover impacts on streamflow: a change-detection modelling approach that incorporates parameter uncertainty. Hydrological Sciences Journal-Journal Des Sciences Hydrologiques 55: 316-332.
- Seibert, J. and McGlynn, B. L. 2007. A new triangular multiple flow direction algorithm for computing upslope areas from gridded digital elevation models. Water Resources Research 43: W04501.
- Sidle, R. C., Ziegler, A. D., Negishi, J. N., Nik, A. R., Siew, R. and Turkelboom, F. 2006. Erosion processes in steep terrain Truths, myths, and uncertainties related to forest management in Southeast Asia. Forest Ecology and Management 224: 199-225.
- Sidle, R. C., Sasaki, S., Otsuki, M., Noguchi, S. and Nik, A. R. 2004. Sediment pathways in a tropical forest: effects of logging roads and skid trails. Hydrological Processes 18: 703-720.
- Singh, V. P. and Woolhiser, D. A. 2002. Mathematical modeling of watershed hydrology. Journal of Hydrological Engineering 7: 270-292.
- Sivapalan, M. 2003. Process complexity at hillslope scale, process simplicity at the watershed scale: is there a connection? Hydrological Processes 17: 1037-1041.
- Sivapalan, M. 2009. The secret to 'doing better hydrological science': change the question! Hydrological Processes 23: 1391-1396.
- Sivapalan, M., Takeuchi, K., Franks, S. W., Gupta, V. K., Karambiri, H, Lakshmi, V., Liang, X., McDonnell, J. J., Mendiondo, E. M., O'Connell, P. E., Oki, T., Pomeroy, J. W., Schertzer, D., Uhlenbrook, S. and Zehe, E.
 2003. IAHS Decade on Predictions in Ungauged Basins (PUB), 2003-2012: shaping an exciting future for the hydrological sciences. Hydrological Sciences 48: 857-880.
- Smerdon, B. D., Redding, T. E. and Beckers, J. 2009. An overview of the effects of forest management on groundwater hydrology. BC Journal of Ecosystems and Management 10: 22-44.
- Smith, L. C. 1997. Satellite remote sensing of river inundation area, stage, and discharge: A review. Hydrological Processes 11: 1427-1439.

- **Sorensen, R. and Seibert, J. 2007**. Effects of DEM resolution on the calculation of topographical indices: TWI and its components. Journal of Hydrology 347: 79-89.
- Sorensen, R., Zinko, U. and Seibert, J. 2006. On the calculation of the topographic wetness index: evaluation of different methods based on field observations. Hydrology and Earth System Sciences 10: 101-112.
- Spatial Planning Systems. 2008. PATCHWORKS. [on-line] www.spatial.ca/products/index.html Deep River, ON.
- Spence, C. and Woo, M-K. 2003. Hydrology of subarctic Canadian shield: soil-filled valleys. Journal of Hydrology 279: 151-166.
- **Stednick, J. D. 1996**. Monitoring the effects of timber harvest on annual water yield. Journal of Hydrology 176: 79-95.
- Stewart, J. B. and Finch, J. W. 1993. Application of remote-sensing to forest hydrology. Journal of Hydrology 150: 701-716.
- Stieglitz, M., Shaman, J., McNamara, J., Engel, V., Shanley, J. and Kling, G. W. 2003. An approach to understanding hydrological connectivity on the hillslope and the implications for nutrient transport. Global Biogeochemical Cycles 17: 1105.
- Stottlemyer, R. and Troendle, C. A. 1999. Effect of subalpine canopy removal on snowpack, soil solution, and nutrient export, Fraser Experimental Forest, CO. Hydrological Processes 13: 2287-2299.
- Summerell, G. K., Vaze, J., Tuteja, N. K., Grayson, R. B., Beale, G. and Dowling, T. I. 2005. Delineating the major landforms of catchments using an objective hydrological terrain analysis method. Water Resources Research 41: 1105.
- Sun, G., McNulty, S. G., Shepard, J. P., Amatya, D. M., Riekerk, H., Comerford, N. B. Skaggs, W. and Swift Jr., L.
 2001. Effects of timber managements on the hydrology of wetland forests in the souther United States. Forest Ecology Management 143: 227-236
- Swanson, R. H. 1998. Forest hydrology issues for the 21st century: A consultant's viewpoint. Journal of the American Water Resources Association 34: 755-763.
- Szaro, R. C. and Peterson, C. E. 2004. Evolving approaches toward science-based forest management. Forest Snow Landscape Research 78: 9-20.
- Tague, C. L. and Band, L. E. 2001a. Evaluating explicit and implicit routing for watershed hydro-ecological models of forest hydrology at the small catchment scale. Hydrological Processes 15: 1415-1439.
- Tague, C. L. and Band, L. E. 2001b. Simulating the impact of road construction and forest harvesting on hydrological response. Earth Surface Processes and Landforms 26: 135-151.
- **Tague, C. L. and Band, L. E. 2004**. RHESSys: Regional Hydro-Ecologic Simulation System-An Object-Oriented Approach to Spatially Distributed Modeling of Carbon, Water, and Nutrient Cycling. Earth Interactions 8: 1-19.
- Tang, Q. H., Gao, H. L., Lu, H. and Lettenmaier, D. P. 2009. Remote sensing: hydrology. Progress in Physical Geography 33: 490-509.
- Tarboton, D. G. 1997. A new method for the determination of flow directions and upslope areas in grid digital elevation models. Water Resources Research 33: 309-319.
- **Tetzlaff, D., Seibert, J., McGuire, K. J., Laudon, H., Burn, D. A., Dunn, S. M. and Soulsby, C. 2009**. How does landscape structure influence catchment transit time across different geomorphic provinces? Hydrological Processes 23: 945-953.

- Thomas, R. B. and Megahan, W. F. 1998. Peak flow responses to clear-cutting and roads in small and large basins, western Cascades, Oregon: a second opinion. Water Resources Research 34: 3393-3403.
- **Thompson, J. C. and Moore R. D. 1996.** Relations between topography and water table depth in a shallow forest soil. Hydrological Processes 10: 1513-1525.
- **Townsend, P. A. 2001**. Mapping seasonal flooding in forested wetlands using multi-temporal radarsat SAR. Photogrammetric Engineering and Remote Sensing 67: 857-864.
- **Troendle, C. A. and Meiman, J. R. 1984**. Options for harvesting timber to control snowpack accumulation. Proceedings of the Western Snow Conference 52: 86-97.
- Tromp-van Meerveld, H. J. and McDonnell, J. J. 2006. Threshold relations in subsurface stormflow: 1. A 147-storm analysis of the Panola hillslope. Water Resources Research 42: W02411.
- US EPA. 1980. An approach to water resources evaluation of non-point silvicultural sources (a procedural handbook). U.S. Environmental Protection Agency, Environmental Research Laborator, Athens, Georgia, EPA-600/3-84-066, 767 pp.
- Vanclay J. K. 2009. Managing water use from forest plantations. Forest Ecology and Management 257: 385-389.
- van der Sanden, J. J. 2004. Anticipated applications potential of RADARSAT-2 data. Canadian Journal of Remote Sensing 30: 369-379.
- VanderKwaak, J. E. 2000. Numerical simulation of flow and chemical transport in integrated surface-subsurface hydrological systems. Dissertation Abstracts International Part B: Science and Engineering 60: 3170.
- Van Dijk, A. I. J., van Noordwijk, M., Calder, I. R., Bruijnzeel, S. L. A., Schellekens, J. and Chappell, N. 2009. Forest-flood relation still tenuous – comment on "Global evidence that deforestation amplifies flood risk and severity in the developing world" by C. J. A. Bradshaw, N. S. Sodi, K. S.-H. Peh and B. W. Brook. Global Change Biology 15: 110-115.
- Van Miegroet, H. and Johnson, D. W. 2009. Feedbacks and synergism among biogeochemistry, basic ecology, and forest soil science. Forest Ecology and Management 258: 2214-2223.
- Vereecken, H., Huisman, J. A., Bogena, H., Vanderborght, J., Vrugt, J. A. and Hopmans, J. W. 2008. On the value of soil moisture measurements in vadose zone hydrology: A review. Water Resources Research 44: W00D06.
- Verhoest, N. E. C., Troch, P. A., Paniconi, C. and De Troch, F. P. 1998. Mapping basin scale variable source areas from multitemporal remotely sensed observations of soil moisture behavior. Water Resources Research 34: 3235-3244.
- Verhoest, N. E. C., Lievens, H., Matgen, P., Montanari, M., Vernieuwe, H., De Lannoy, G. J. M., De Baets, B. and Hoffmann, L. 2008. Soil moisture retrieval from ALOS PALSAR in the Alzette (Luxembourg) and Zwalm (Belgium) catchments. The International Workshop on Microwave Remote Sensing for Land Hydrology Research and Applications. October 20-22, 2008, Oxnard, California, USA.
- Verstraeten, W. W., Veroustraete, F. and Feyen, J. 2008. Assessment of evapotranspiration and soil moisture content across different scales of observation. Sensors 8: 70-117.
- Vitousek, P. M. and Reiners, W. A. 1975. Ecosystem succession and nutrient retention: a hypothesis. Bioscience 25: 376-381.

- Vorosmarty, C. J., Willmott, C. J., Choudhury, B. J., Schloss, A. L., Stearns, T. K., Robeson, S. M. and Dorman, T. J. 1996. Analyzing the discharge regime of a large tropical river through remote sensing, ground-based climatic data, and modeling. Water Resources Research 32: 3137-3150.
- Wagner, W., Bloschl, G., Pampaloni, P., Calvet, J. C., Bizzarri, B., Wigneron, J. P. and Kerr, Y. 2007. Operational readiness of microwave remote sensing of soil moisture for hydrological applications. Nordic Hydrology 38: 1-20.
- Watson, R., Noble, I., Bolin, B., Ravindranath, N. H., Verardo, D. J. and Dokken, D. J. 2000. Summary for policymakers: land-use, land-use change, and forestry. IPCC Special Report. 92-9169-114-3.
- Webster, K. L., Creed, I. F., Bourbonniere, R. A. and Beall, F. D. 2008a. Controls on the heterogeneity of soil respiration in a tolerant hardwood forest. Journal of Geophysical Research-Biogeosciences 113: G03018.
- Webster, K. L., Creed, I. F., Beall, F. D. and Bourbonniere, R. A. 2008b. Sensitivity of catchment-aggregated estimates of soil carbon dioxide efflux to topography under different climatic conditions. Journal of Geophysical Research-Biogeosciences 113: G03040.
- Webster, K. L., Creed, I. F., Beall, F. D. and Bourbonniere, R. A. 2010. A topographic template for estimating soil carbon pools in forested catchments. Geoderma 160: 457-467.
- Webster, R. 1977. Quantitative and numerical methods in soil classification and survey. Oxford: Clarendon Press. 269 pp.
- Wemple, B. C., Jones, J. A., Grant, G. E. 1996. Channel network extension by logging roads in two basins, western Cascades, Oregon. Water Resources Bulletin 32: 1195-1207.
- Westbrook, C., Devito, K. J., Allan C. J. 2006. Soil N cycling in harvested and pristine boreal forests and peatlands. Forest Ecology and Management 234: 227-237.
- Western, A. W., Bloschl, G. and Grayson, R. B. 2001. Toward capturing hydrologically significant connectivity in spatial patterns. Water Resources Research 37: 83-97.
- Whitaker, A., Alila, Y., Beckers, J. and Toews, D. 2003. Application of the distributed hydrology soil vegetation model to redfish creek, British Columbia: model evaluation using internal catchment data. Hydrological Processes 17: 199-224.
- Whitcomb, J., Moghaddam, M., McDonald, K., Kellndorfer, J. and Podest, E. 2009. Mapping vegetated wetlands of Alaska using L-band radar satellite imagery. Canadian Journal of Remote Sensing 35: 54-72.
- Whitehead, P. G. and Robinson, M. 1993. Experimental basin studies an international and historical perspective of forest impacts. Journal of Hydrology 145: 217-230.
- Whitson, I. R., Chanasyk, D. S., Prepas, E. E. 2003. Hydraulic properties of Orthic Gray Luvisolic soils and impact of winter logging. Journal of Environmental Engineering Science 2: S41-S49.
- Wilson, J. and Gallant, J. C. 2000. Terrain Analysis: Principles and Applications. New York: Wiley. 479 pp.
- Winn, N., Williamson, C. E., Abbitt, R., Rose, K., Renwick, W., Henry, M. and Saros, J. 2009. Modeling dissolved organic carbon in subalpine and alpine lakes with GIS and remote sensing. Landscape Ecology 24: 807-816.
- Wolock, D. M. and McCabe, G. J. 1995. Comparison of single and multiple flow direction algorithms for computing topographic parameters in TOPMODEL. Water Resources Research 31: 1315-1324.

- Wood, E. F., Sivapalan, M., Beven, K. and Band, L. 1988. Effects of spatial variability and scale with implications to hydrological modeling. Journal of Hydrology 102: 29-47.
- Wu, W., Hall, C. A. S., Scatena, F. N. and Quackenbush, L. J. 2006. Spatial modelling of evapotranspiration in the Luquillo experimental forest of Puerto Rico using remotely-sensed data. Journal of Hydrology 328: 733-752.
- Yaffee, S. L. 1999. Three faces of ecosystem management. Conservation Biology 13: 713-725.
- Yang, F. H., White, M. A., Michaelis, A. R., Ichii, K., Hashimoto, H., Votava, P., Zhu, A. X. and Nemani, R. R.
 2006. Prediction of continental-scale evapotranspiration by combining MODIS and AmeriFlux data through support vector machine. IEEE Transactions on Geoscience and Remote Sensing 44: 3452-3461.
- Yarrow, M. M. and Marin, V. H. 2007. Toward conceptual cohesiveness: A historical analysis of the theory and utility of ecological boundaries and transition zones. Ecosystems 10: 462-476.
- Young, A. 1972. The soil catena: A systematic approach. In International geography Vol 1, International Geography Congress, W. P. Adams and F. M. Helleiner (eds.). University of Toronto Press, Toronto, 287-289.
- Young, A. 1976. Tropical Soils and Soil Survey. Cambridge, Cambridge University Press. 468 pp.
- Zhang, W. H. and Montgomery, D. R. 1994. Digital elevation model grid size, landscape representation, and hydrological simulations. Water Resources Research 30: 1019-1028.

SFM Network Partners

August 2007

GRANTING COUNCILS

- Networks of Centres of Excellence / Government of Canada
- Natural Sciences and Engineering Research Council of Canada (NSERC)
- Social Sciences and Humanities Research Council of Canada (SSHRC)

PARTNERS

Governments

- Government of Canada (Environment Canada) (Natural Resources Canada, Canadian Forest Service) (Parks Canada, Ecological Integrity Branch)
- Government of Alberta
 (Advanced Education and Technology –
 Alberta Forestry Research Institute)
 (Sustainable Resource Development)
- Government of British Columbia (Ministry of Forests and Range)
- Government of Manitoba (Manitoba Conservation)
- Government of Newfoundland and Labrador (Department of Natural Resources)
- Government of Ontario (Ministry of Natural Resources)
- Government of Québec (Ministère des Ressources naturelles et de la Faune)
- Government of Yukon (Department of Energy, Mines and Resources)

Industries

- Abitibi Bowater Inc.
- •Alberta-Pacific Forest Industries Inc.
- Canadian Forest Products Ltd.
- Daishowa-Marubeni International Ltd.
- J.D. Irving, Limited
- Louisiana-Pacific Canada Ltd.
- Manning Diversified Forest Products Ltd.
- Tolko Industries Ltd.
- Tembec Inc.
- Weyerhaeuser Company Ltd.

NGO

• Ducks Unlimited Canada

Aboriginal Groups

- Gwich'in Renewable Resource Board
- Heart Lake First Nation
- Kamloops Indian Band
- Kaska Tribal Council
- Little Red River Cree Nation
- Métis National Council
- Moose Cree First Nation
- Treaty 8 First Nations of Alberta

Institutions

- University of Alberta (host institution)
- British Columbia Institute of Technology
- Concordia University
- Dalhousie University
- Lakehead University
- McGill University
- Memorial University of Newfoundland
- Mount Royal College
- Royal Roads University
- Ryerson University
- Simon Fraser University
- Thompson Rivers University
- Trent University
- Université de Moncton
- Université de Montréal
- Université de Sherbrooke
- Université du Québec à Chicoutimi
- Université du Québec à Montréal
- Université du Québec à Rimouski
- Université du Québec à Trois-Rivières
- Université du Québec en Abitibi-Témiscamingue
- Université Laval
- University of British Columbia
- University of Calgary
- University of Guelph
- University of Lethbridge
- University of Manitoba
- University of New Brunswick
- University of Northern British Columbia
- University of Ottawa
- University of Regina
- University of Saskatchewan
- University of Toronto
- University of Victoria
- University of Waterloo
- University of Western Ontario
- University of Winnipeg
- Wilfrid Laurier University

Affiliated Members

- Canadian Institute of Forestry
- Forest Ecosystem Science Cooperative, Inc.
- Forest Engineering Research Institute of Canada (FERIC)
- Fundy Model Forest
- Lake Abitibi Model Forest
- Manitoba Model Forest
- National Aboriginal Forestry Association



Sustainable Forest Management Network www.sfmnetwork.ca

Networks of Centres of Excellence

